**Collective Bargaining Agreement** 

between

# **Campbell Union School District**

and

# Campbell Elementary Teachers Association, Affiliate of California Teachers Association and National Education Association

# 2022-2025

Tentative Agreement signed: September 11, 2023

Ratified by the Governing Board: September 28, 2023

Effective: July 1, 2022 - June 30, 2025

ARTIC	LE 1 - PREAMBLE AND RECOGNITION	
1.1	Preamble	1
1.2	Recognition	1
1.3	Job Descriptions	1
1.4	New Certificated Positions	
ARTICI	LE 2 - NON-DISCRIMINATION	2
ARTIC	LE 3 - DISTRICT RIGHTS	
3.1	Description of Rights	
3.2	Limitations of Rights	
3.3	Declaration of Emergency	
3.4	Arbitrability	
ARTICI 4.1	LE 4 - ASSOCIATION RIGHTS	
4.2	Association Leave	
4.3	Dues Deduction	
4.4	Service Fee and Maintenance of Membership	
4.5	Forwarding of Dues Deducted	
ARTIC	LE 5 - GRIEVANCE PROCEDURE	
5.1	Definitions	
5.2	Purpose	
5.3	Procedures	
5.4	Miscellaneous Provisions	
ARTIC	LE 6 - CLASS SIZE	17
6.1	Staffing Ratio	

6.2	Equalizing Teaching Loads	
6.3	Class Size Support for TK	
6.4	Class Size Overages for K-3	
6.5	Appeal to Building Advisory Council or Committee	
6.6	Lab Classes and PE Classes	
ARTICI	LE 7 - EVALUATION PROCEDURES	20
7.1	Frequency of Evaluation	
7.2	Areas of Evaluation	
7.3	Use of Publishers' Norms	21
7.4	Notice to Unit Members	21
7.5	Pre-Conference and Evaluation Plan	21
7.6	Classroom Observation/Post Observation Conference	
7.7	Formal Evaluation Summary	
7.8	Personnel Files Relating to Evaluations	23
7.9	Complaint Procedure	24
7.10	Status in Position	25
7.11	Alternative Professional Growth Evaluation Plans	25
ARTICI	LE 8 - ASSIGNMENTS, REASSIGNMENTS AND TRANSFERS	27
8.1	Assignments and Change of Assignments	27
8.2	Reassignment Due to Layoff	
8.3	Unit Member-Initiated (Voluntary) Transfers	
8.4	Involuntary Transfer Procedures	30
8.5	Relocation Assistance	
ARTICI	LE 9 - HOURS AND DAYS OF EMPLOYMENT	32
9.1	Professional Day	
9.2	Instructional Minutes	
9.3	Adjunct Duties	

9.4	Mandatory Meetings	
9.5	Coordination with CETA Meetings	
9.6	Night Meetings	
9.7	Duty-Free Lunch	
9.8	Unit Member Preparation Time	
9.9	Shortened Student Instructional Days	
9.10	Parent/Teacher Intervention Conferences	
9.11	Length of Work Year	
9.12	Election Day Meetings	
ARTICL	E 10 - COMPENSATION	
10.1	Salary Schedule	
10.2	Salary of Employees Not Completing Year of Employment	
10.3	Salary Warrants	
10.4	Repair or Replacement of Personal Property	
10.5	Health and Welfare Benefits	
10.6	Extended Duty Pay	
10.7	Pay for NBPTS Certification	
10.8	Stipends	
10.9	Teacher Travel and Overnight Supervision	
10.10	Substitute for Absent Teacher	
10.11	Psychologists and Nurses	
10.12	STRS Defined Supplemental Benefits Plan	
ARTICL	E 11 - SAFETY CONDITIONS	50
11.1	Employee Protection	
11.2	Working Conditions	
11.3	Suspension	51
11.4	Support Services	51

11.5	Disclosure	52
ARTICL	E 12 - SUMMER SCHOOL	53
12.1	Employment	53
12.2	Compensation	53
12.3	Sick Leave	53
ARTICI	E 13 - PARTIAL CONTRACTS, JOB SHARING, AND PART-TIME HOURLY	54
13.1	Partial Contracts	54
13.2	Job Sharing	54
13.3	Part-Time Hourly	55
ARTICL	E 14 - LEAVES	56
14.1	Personal Illness and Injury	56
14.2	Personal Necessity Leave	58
14.3	Bereavement Leave	59
14.4	Leave for Pregnancy Disability	59
14.5	Leave for Bonding with Recently Placed Foster or Adopted Child or Newborn Infant	60
14.6	Leave Without Pay for Child Bearing Preparation	60
14.7	Leave Without Pay for Child Rearing	61
14.8	Industrial Accident Leave	62
14.9	Judicial Leave	63
14.10	Legislative Leave	63
14.11	Military Leave	63
14.12	Family and Medical Leave Act	64
14.13	Other Leaves Without Pay	64
14.14	Catastrophic Leave Bank	64
14.15	Miscellaneous	70

ARTICL	E 15 - ACADEMIC FREEDOM	71
15.1	Academic Freedom	71
15.2	Board Policy	
ARTICL	E 16 - TEACHER RIGHTS TO DUE PROCESS	72
16.1	General Provisions	72
16.2	Progressive Discipline	72
16.3	Discipline Without Progression	73
16.4	Procedures — Public Charges	73
16.5	Administrative Leave With Pay	73
16.6	Notice of Discipline	73
16.7	Appeal of Discipline	73
16.8	Arbitration	74
16.9	Association Presence	74
16.10	Effect of Pending Appeal	74
16.11	Confidentiality	74
16.12	Limitations	75
16.13	Post Discipline	75
16.14	Application	75
ARTICL	E 17 - RETIREMENT PROVISIONS	76
17.1	Pre-Retirement/Part-Time Employment	
17.2	Early Retirement/Consultant Contract	
17.3	Retiree Benefits	
17.4	Retirement for District Determined Disability	79
17.5	Post Retirement Employment Program	
17.6	Concurrent Participation	

# 

18.1	SBSRDM	
18.2	Procedure	
18.3	Waiver Approval	82
18.4	Record of Waivers	
		00
ARTI	CLE 19 - SPECIAL EDUCATION	83
19	.1 Release days for impactful caseloads	83
19	.2 Non-work days	83
19	.3 Prep time	83
19	.4 SDC support for more than 3 grade levels	83
ARTI	CLE 20 - MISCELLANEOUS PROVISIONS	84
20.1	Effects of agreement	84
20.2	Individual Contracts	84
20.3	Savings	84
20.4	Support of Agreement	84
20.5	Concerted Activities	85
20.6	Term of Agreement	85
20.7	Printing the Agreement	86

APPENDIX A - SALARY SCHEDULE	87
APPENDIX B – CALENDAR	
APPENDIX C - PEER ASSISTANCE AND REVIEW PROGRAM	
APPENDIX D - EVALUATION FORMS	
APPENDIX E - SHARED TEACHING CONTRACT GUIDELINES	

# **ARTICLE 1 - PREAMBLE AND RECOGNITION**

# 1.1 Preamble

This Agreement is entered into this 1st day of July 2022, between the Governing Board of the Campbell Union School District ("District") and the Campbell Elementary Teachers Association ("Association"), an affiliate of the California Teachers Association, and the National Education Association.

## 1.2 Recognition

The Board recognizes this Association as the exclusive representative of all regular, nonadministrative certificated employees of the Board, excluding substitute teachers and the following management and supervisory personnel:

Superintendents	Principals
Directors	Assistant Principals
Coordinators	Management Interns

# **1.3** Job Descriptions

Prior to instituting any changes in the bargaining unit job descriptions, the District will consult with the Association regarding such changes.

## **1.4** New Certificated Positions

New certificated positions which are established during the term of this Agreement shall first be reviewed by the District and the Association as to their inclusion in the bargaining unit, and shall thereafter be accredited to the unit if the positions share a community of interest with the existing unit. In the event the parties fail to agree on the inclusion or exclusion of the positions, the dispute will be referred to the Public Employment Relations Board for a decision.

# **ARTICLE 2 - NON-DISCRIMINATION**

- 2.1 All Articles and provisions of this Agreement shall be applied equitably to all unit members.
- 2.2 This Agreement shall not be applied in an arbitrary, discriminatory or capricious manner.
- 2.3 No unit member shall be treated discriminatorily based on race, color, creed, age, sex, national origin, marital status, sexual orientation, a physical disability, religion, membership in an employee organization or participation in an employee organization.

#### **ARTICLE 3 - DISTRICT RIGHTS**

# **3.1** Description of Rights

It is understood and agreed that the District retains all of its powers and authority to direct, manage and control to the full extent of the law. Included in but not limited to those duties and powers are the exclusive rights to: determine its organization; direct the work of its unit members; determine the times of operation; determine the kinds and levels of services to be provided, and the methods and means of providing them; establish its educational policies, goals and objectives; insure the rights and educational opportunities of students; determine staffing patterns; determine the number and kinds of personnel required; maintain the efficiency of District operations; determine the curriculum; build, move or modify facilities; establish budget procedures and determine budgetary allocation; determine the methods of raising revenue; contract out work; and take action on matters in the event of an emergency. In addition, the Board retains the right to hire, classify, assign, transfer, evaluate, promote, terminate, and discipline unit members.

# **3.2** Limitations of Rights

The District's exercise of the foregoing powers, rights, authority, duties, and responsibilities, the adoption of policies, rules, regulations and practices in furtherance thereof, and the use of judgment and discretion in connection therewith, shall be limited by the terms and conditions of this Agreement, and then only to the extent such terms and conditions are in conformance with law.

# **3.3** Declaration of Emergency

The declaration of an emergency shall be limited to natural disasters and/or events over which the District has no control. In no event shall the District declare an emergency in order to evade the provisions of this Agreement.

#### 3.4 Arbitrability

The exercise of the discretionary powers reserved to the District by this provision shall not be subject to review by an arbitrator nor subject to the grievance procedure, except that an arbitrator shall have the authority to determine the applicability of this article to any dispute regarding the violation of application of this Agreement.

#### **ARTICLE 4 - ASSOCIATION RIGHTS**

# 4.1 Unit Member Communications and Meetings

- 4.1.1 The Association shall use the school mailboxes and bulletin board spaces provided that it gives the District and principals copies of general mailings and announcements. All items for bulletin boards must contain the date of posting, the identification of the Association and shall not be obscene or defamatory. All postings on bulletin boards shall be done by an authorized Association representative. It is understood that any member of CETA is eligible to be an authorized representative. The Association shall provide the District with a list of authorized representatives as soon as possible.
- 4.1.2 The Association may use District mail services to the extent permitted by law.
- 4.1.3 The names of all unit members listed by location shall be provided to the Association without cost as soon as possible after the beginning of the school year. Upon the Association's request, a second updated list shall be provided to the Association prior to the end of each February.
- 4.1.4 Prior to the first working day each year, the District shall provide the Association with the names and addresses of all the new unit members who were employed during the summer months, unless the new unit members request, in writing, that their addresses not be provided to the Association.
- 4.1.5 The District directory shall be provided without cost to the Association as soon as possible after printing.
- 4.1.6 During the District's business hours, the District shall provide the Association with access to information which is necessary in order to carry out its duties and responsibilities under the Public Employment Relations Act. Upon request, the District will provide a copy of the requested material. In extenuating circumstances where the material requested is voluminous, the District will allow the Association access to the material, which may then be copied by an authorized Association representative.

- 4.1.7 The Association shall have the right to use school meeting rooms and conduct Association business subject to availability of space as determined by the site administrator, and provided the Association business does not interfere with the unit members' obligation to the District.
- 4.1.8 A list of the names of unit members who have been served layoff notices recommending non-reemployment under Education Code Sections 44949 and 44955 shall be sent to the Association within twenty-four (24) hours of the time of service upon the affected unit member.

# 4.2 Association Leave

4.2.1 Provided advance notice of at least two (2) days is given the District, the Association may designate such members, as it deems necessary, who are Association representatives, to receive paid leaves of absence. The total number of leaves will not exceed fifteen (15) full workdays in each full school year. Additional leave may be granted at the Superintendent's discretion. Leaves of less than one (1) full day may be granted if mutually agreed upon.

4.2.2 The Association President shall be granted paid release <u>up to 100%</u> time of his/her full-time equivalent assignment for each year of this Agreement to conduct Association business. The Association shall determine the percentage of the release time up to one hundred percent 100%. The Association shall notify the District of the percentage release no later than March 15. The Association shall reimburse the District at the salary schedule placement of Step 9, Column II for the percentage of full-time equivalent release time that the President uses. While on release time as defined by this section, the Association President shall be entitled to the benefits created by Education Code Section 22711 and 44987.

6

# 4.3 **Dues Deduction**

4.3.1 Unit members shall be provided the opportunity for payroll deductions designated for NEA-Fund and ABC.

# 4.4 Service Fee and Maintenance of Membership

4.4.1 <u>Employee Rights</u>

The District and CETA/CTA/NEA recognize the right of employees to form, join and participate in lawful activities of employee organizations and the equal, alternative right of employees to refuse to form, join and participate in employee organizations. Neither party shall exert pressure upon nor discriminate against an employee in the exercise of these alternative rights. Accordingly, membership in the Association shall not be compulsory. An employee has the right to choose to be an active member of CETA/CTA/NEA

- 4.4.2 <u>Payroll Deductions</u>
  - 4.4.2.1 The Association shall be provided continuous payroll deductions of unified dues of active members. The Association shall hold the District harmless on account of any problem arising from monies having been deducted and remitted to the Association pursuant to this Section. The Association agrees to furnish any information needed by the District to fulfill the provisions of this Section.
  - 4.4.2.2 The Association is entitled to a change in payroll deductions of its member(s) provided an authorized Association officer submits a written request to the District for such adjustment; and provided further that at least thirty (30) calendar days prior to the change an authorized Association officer shall furnish the District with evidence that the Association provided notification to its members of the change. Such changes shall consist of total annual dues change, increase or decrease in dues requirements, and increase or decrease in authorized deductions by a member(s) for Association approved items.

- 4.4.2.3 The District shall remit all dues collected to the Association with an accompanying alphabetical list of all unit members for whom all dues deductions have been made.
- 4.4.2.4 Any Agency Fee payer under the superseded version of this Article, as well as any potential member who does not make application for membership upon being hired by CUSD, shall not have membership or agency fee dues deducted from his/her pay warrant.
- 4.4.2.5 The parties agree that membership adding new members, maintaining current members, or dropping members who complete the process with CETA to do so is entirely a function of the union, and no part of the Collective Bargaining Agreement may interfere with the union membership process.
- 4.4.2.6 The District shall not end dues deductions for a member who wishes to terminate his/her membership until the CETA/CTA/NEA notifies the District that the request to drop membership has been processed by the CTA.
- 4.4.2.7 (1) The District is under no obligation to make payroll deductions for periods during which a unit member is either terminated from active employment or not on the District's active payroll for any reason, including, but not limited to, layoff and voluntary leave of absence for more than thirty (30) days.
  - (2) The unit member's earnings must be sufficient after other legal and required deductions are made to cover the amount of the dues or service fees to be deducted.
  - (3) When a unit member is in a non-pay status for an entire pay period, no withholding will be made to cover the pay period. In the case of a unit member who is in a non-pay status during only part of the pay period, and the salary is not sufficient to cover the full

8

withholding, no deduction shall be made. All other legal and required deductions have priority over Association dues and service fees.

#### 4.4.3 Hold Harmless and Indemnity Provision

- 4.4.3.1 The California Teachers Association, as defined by this Agreement, shall hold the District harmless and shall fully and promptly reimburse the District for reasonable legal fees and costs incurred in responding to or defending against any claims or disputes which are actually filed against the District or any of its agents by other than the Association in connection with the administration or enforcement of any section of this Agreement pertaining to representation fee. Reimbursement shall include costs and attorney's fees incurred by the District.
- 4.4.3.2 Upon notice that the District is going to seek indemnification or to be held harmless under this provision, the Association shall have the right to meet with the District regarding the reasonableness and merit of any claim, demand, suit or action for which the District seeks indemnification, and shall attempt to agree whether any action listed above in Section 4.4.8.1 shall be compromised, resisted, defended, tried or appealed.
- 4.4.3.3 In determining whether or not any action listed in Section 4.4.8.1 shall be compromised, resisted, defended, tried or appealed, the District will defer to the Association's interest if the District does not have a distinct and separate legal interest in the matter in dispute.
- 4.4.3.4 The District shall not be entitled to be reimbursed for any fees, costs, charges or penalties for which the Association was not properly notified and provided the opportunity to discuss as set forth in this Article; nor will the District be entitled to any reimbursement when the District's efforts in defending against an action would be duplicative, or when the District is

defending a separate and distinct legal interest or when the District is defending an activity which is arguably subject to criminal liability on the part of any District administrator.

# 4.5 Forwarding of Dues Deducted

With respect to all sums deducted by the District pursuant to unit member authorization, the District agrees to promptly remit such monies to the California Teachers Association, accompanied by an alphabetical list of names of unit members for whom deductions have been made.

#### **ARTICLE 5 - GRIEVANCE PROCEDURE**

#### 5.1 **Definitions**

- 5.1.1 A "grievance" is a formal written allegation that there has been a violation, misinterpretation or misapplication of specific provision(s) of this Agreement.
- 5.1.2 A "day" for this Article shall be any day in which the grievant is required to render service to the District.
- 5.1.3 The appropriate designated District representative shall be the management person most familiar with the situation, and who has been designated by the District to adjust grievances. A member shall be notified if the immediate administrator is someone other than the management person having immediate jurisdiction over the member.
- 5.1.4 The grievant's representative shall be appointed by the Association, if the grievant wishes to be represented.
- 5.1.5 A "grievant" is any member(s) of the bargaining unit, or Association, asserting a grievance.

# 5.2 Purpose

The purpose of the grievance procedure is to attempt to secure equitable solutions to grievances at the lowest possible level. The grievant and immediate supervisor or the administrator most familiar with the circumstances of the grievance should attempt to resolve the grievance at the informal level.

# 5.3 Procedures

#### 5.3.1 <u>Informal Resolution</u>

5.3.1.1 Any grievant shall attempt to present the grievance orally to their immediate supervisor or the administrator most familiar with the circumstances of the grievance within fifteen (15) days after the grievant knew, or reasonably should have known of the circumstances which form the basis for the grievance. The immediate supervisor or administrator shall meet with the unit member and attempt to resolve the matter within ten (10) days after the presentation of the grievance.

- 5.3.1.2 The parties to the grievance may be represented at this conference. The grievant may request an additional five (5) days extension in writing from the Human Resources Office. The extension will be granted if the written request is received by the Human Resources Office prior to the expiration of the time limit in Section 5.3.1.1.
- 5.3.1.3 Association Grievance: A grievance initiated by the Association may be filed at Level Two subject to compliance with the timelines set forth in 5.3.1.1.
- 5.3.2 Level One: Formal Written Grievance
  - 5.3.2.1 If the grievance has not been resolved informally, the grievant may present a written grievance on the appropriate form within forty-five (45) days of the act or omissions which gave rise to the grievance, or within forty-five (45) days of when the act or omissions first came to the grievant's attention; or in the exercise of reasonable diligence should have come to the grievant's attention or within 20 days following the informal denial. The Level One grievance will be filed with the supervisor or administrator who participated in the informal resolution meeting.
  - 5.3.2.2 The written statement of the grievance shall contain:
    - A description of the specific grounds for the grievance, including names, dates, and places necessary for a complete understanding of the grievance;
    - (2) A listing of the provisions of this Agreement which are alleged to have been violated;
    - (3) A listing of the reasons the immediate supervisor's proposed resolution of the problem is unacceptable; and
    - (4) A listing of specific actions requested of the District which will remedy the grievance.

- 5.3.2.3 The District's representative shall communicate a written decision and the reasons for the decision to the grievant within ten (10) days after receiving the grievance. If the District's representative does not respond within the time limits, the grievant may proceed to the next level. Within the above time limits, either party may request a personal conference.
- 5.3.3 Level Two: Appeal to Superintendent
  - 5.3.3.1 If the grievant is not satisfied with the decision at Level One, or if there has been no response at Level One within twenty (20) days of submission of the formal written grievance, the grievant may appeal the decision in writing to the Superintendent or designee. The designee will not be the same supervisor or administrator or administrator who responded at Level One. If the grievant does not appeal in writing within ten (10) days of receipt of the Level One response, the grievance is deemed to be resolved.
  - 5.3.3.2 The written appeal shall include a copy of the original grievance, the decision rendered, and a statement of the reason(s) for the appeal.
  - 5.3.3.3 The Superintendent or designee shall conduct an investigation and, upon request, hold a conference with the grievant. The parties to the grievance may be represented at this conference. Within ten (10) days of the receipt of the appeal or any amendment of the appeal, the Superintendent or designee shall deliver to the grievant a written decision, with the reasons for the decision.
  - 5.3.3.4 If the Superintendent or designee does not respond within the ten (10) day time limitation, the Association may proceed to the next level.
- 5.3.4 Level Three: Binding Arbitration
  - 5.3.4.1 If not satisfied with the decision at Level Two, the Association may submit a written demand for arbitration to the Superintendent. Only issues which were processed and handled in accordance with the grievance procedure of this Article are subject to arbitration. The grievance is resolved if a

written request for arbitration is not submitted within twenty (20) days after the grievant receives the

Superintendent's decision, or if there has been no response at Level Two, within twenty (20) days of the deadline for receipt of the Level Three response.

- 5.3.4.2 Either party may request that the American Arbitration Association (AAA) supply a panel of seven (7) names of arbitrators experienced in public sector grievances. The selection of an arbitrator shall be made in accordance with the Voluntary Labor Arbitration Rules of the American Arbitration Association, and the parties shall be bound by the Voluntary Labor Arbitration Rules of the AAA.
- 5.3.4.3 Within seven (7) days of the selection of the arbitrator, the Superintendent and the grievant or designees shall attempt to agree in writing upon the issue or issues to be submitted to the arbitrator. If they are unable to agree upon a submission statement, the arbitrator shall determine the issues by referring to the written grievance and answers to the grievance at each level.
- 5.3.4.4 The arbitrator shall render a written decision on the submitted issue(s) in accordance with AAA rules after the close of the hearing, or if an oral hearing has been waived, after the final submission of written evidence and final arguments.
- 5.3.4.5 The District and the Association agree that the jurisdiction and authority of the arbitrator so selected in opinions he/she expresses will be confined exclusively to the interpretation of the expressed provision or provisions of the Agreement at issue between the parties. The arbitrator shall have no authority to add to or subtract from, alter, amend, or modify the provisions of this Agreement. The arbitrator's decision within the limits prescribed shall be final and binding upon the parties to the dispute.

5.3.4.6 The fees and expenses of the arbitrator and the hearing shall be borne equally by the District and the grievant. All other expenses shall be borne by the parties incurring them. Unless the parties mutually agree to share the expenses, the cost of the services and expenses of the court reporter shall be paid by the party requesting same. If however, one of the parties declines to share the expenses of the court reporter and subsequently requests a copy of the transcript, that party shall be required to reimburse the other party one-half (1/2) of the cost of the court reporter's services and expenses. The cost of any transcript or any copy of any transcript requested by either party shall be borne by the party requesting same.

# 5.4 Miscellaneous Provisions

- 5.4.1 Relevant unit member witnesses, the grievant and the grievant's representative shall be provided release time without loss of pay for the purpose of participating at grievance arbitration hearing(s). Witnesses may be placed on on-call status and required to attend the arbitration hearing for only so long as his/her testimony is taken. By mutual agreement, relevant unit member witnesses, the grievant and the grievant's representative shall be provided release time without loss of pay for the purpose of preparing for the arbitration hearing.
- 5.4.2 All documents, communications and records dealing with the processing of a grievance will be filed separately from the participant's personnel file and maintained in the Superintendent's office.
- 5.4.3 No party to a grievance shall take any reprisals against the other party to the grievance because he/she participated in an orderly manner in the grievance procedure.
- 5.4.4 If two or more bargaining unit members have the same grievance in issue and fact, then the grievance shall be consolidated for purposes of hearing and decision; provided further, that the rights of the parties are not prejudiced by the consolidation.
- 5.4.5 Time limits contained in this Article may only be waived by mutual agreement.

15

- 5.4.6 In the event a grievance is filed at such time that it cannot be processed through all levels of procedure by the last working day of the school year, the time limits set forth in this Article will be reduced upon mutual agreement, so that the procedure may be completed prior to the end of the school year, or as soon thereafter as it is practicable.
- 5.4.7 No grievance shall be arbitrated without the Association's consent.

#### **ARTICLE 6 - CLASS SIZE**

# 6.1 Staffing Ratio

The assignment ratio of grade 4-8 students to classroom teachers shall be thirty (30) students to one (1) teacher, based on school site average. The District may staff TK-3 classes up to twenty-four (24) to one (1) teacher, based on site average. Special day students mainstreamed into elementary classrooms thirty-four percent (34%) or more of the day shall be counted in the staffing ratio.

6.1.1 Should class size for a "Special Day" class exceed twelve (12) students, and/or students represent three (3) or more grade levels, the Special Education Director and the member shall meet to determine what shall be put in place to support the member. Supports may include, but are not limited to educational aide support, release time, training or coaching support. The member may bring a representative or colleague to the meeting.

6.1.2 Special Education Teachers shall not be counted in the site staffing ratio.

# 6.2 Equalizing Teaching Loads

The District shall attempt to equalize the teaching load for all classroom teachers within similar disciplines and responsibility levels. The process of equalizing class size will be initiated within two (2) weeks of school opening and the District will make reasonable efforts to have the initial equalization process completed by September 20. In the event that equalization is still in process on September 20, the principal will meet with the affected members to provide an update. When setting up classes and equalizing the teaching load, consideration will be given to the teacher serving students with exceptional needs.

### 6.3 Class Size Support for TK

Class size shall reflect new State Guidelines. In the event that the State staffing guidelines of students to adults allows more students than the current 24:2 ratio, the following will apply: If a Transitional Kindergarten class has 20-23 students, the District will provide a half day Aide support for the classroom. If a Transitional Kindergarten class has 24 or more students, the District will provide a full day Aide support for that classroom.

#### 6.4 Class Size Overages for K-3

Retroactive to Day 11, K-3 teachers will be compensated \$7.50 (seven dollars and fifty cents) per student per day for each day that class size exceeds the assignment ratio in Article 6.1. Payment shall be made in the February paycheck for Day 11– January 31 and in the June paycheck for February 1– May 31.

## 6.5 Appeal to Building Advisory Council or Committee

Any unit member having a teaching load which is disproportionate to teaching loads and responsibilities of other teachers of the same type or grade level may request that the Building Advisory Council, established in Section 9.3 of this Agreement, investigate the situation. If the situation is not resolved at the Building Advisory Council level, it will be appealed to the Standing District Advisory Council, which will make written recommendations to the Superintendent in an attempt to resolve the problem. The Standing District Advisory Council shall be composed of three (3) members appointed by the Association and three (3) members appointed by the Superintendent or designee.

- 6.5.1. If a 4-8 class size reaches 34 in math, science, language arts and or social studies on or after September 20, the classroom teacher may appeal to the Principal. If the classroom teacher is dissatisfied with the Principal's response, he/she shall have the right to appeal to the Superintendent or designee.
- 6.6 Lab Classes and PE Classes
  - 6.6.1 No middle school lab class shall contain more students than the number of student work stations. If a lab classroom must be used for non-lab classroom purposes, this section shall not apply.
  - 6.6.2. If a PE class exceeds 38 students, PE teachers will be compensated \$7.50 (seven dollars and fifty cents) per student per period per day for each day that class size exceeds 38 students. Payment shall be made in the February paycheck for Day 11-January 31 and in the June paycheck for February 1-May 31.
  - 6.6.3 By the first instructional day of the school year, all PE teachers will be assigned and have continuous access to a classroom equivalent in size and student capacity to that

of other classroom teachers at that school site. At no time shall a PE class size ever exceed the number of student desks that can reasonably fit in that classroom.

#### **ARTICLE 7 - EVALUATION PROCEDURES**

#### 7.1 Frequency of Evaluation

#### 7.1.1 <u>Probationary/Temporary Unit Members</u>

Each probationary and/or temporary unit member shall be evaluated on a continuing basis. Each probationary and/or temporary unit member shall be given one interim informal evaluation report on or before November 7 and a final evaluation report on or before February 15.

# 7.1.2 Permanent Unit Members

Each permanent unit member shall be evaluated on a continuing basis. Permanent unit members meeting standards shall be formally evaluated once every other year. Permanent unit members shall be provided with an interim report following the first formal observation. The Permanent Interim Evaluation Summary form shall be used unless, by mutual agreement, another form of written feedback is used. Permanent unit members given a rating of "Progress Towards Standards Not Evident" in any category on the Formal Evaluation Summary shall be formally evaluated each school year until successful completion of the process described in section 7.7.2 or suspension/termination as described in section 7.7.3.

Permanent unit members, who have been employed at least 10 years with the district, are highly qualified, and whose previous evaluation rates the employee as meeting standards, may be formally evaluated once every five (5) years.

# 7.2 Areas of Evaluation

The District shall evaluate and assess certificated unit member performance as it reasonably relates to professional standards. Classroom teachers will be evaluated on their implementation of the California Standards for the Teaching Profession (CSTP), as noted in Appendix D. Non classroom bargaining unit members will be evaluated using the evaluation templates for the various roles and responsibilities, as noted in Appendix D.

# 7.3 Use of Publishers' Norms

The evaluation and assessment of unit members' performance pursuant to this Article shall not include the use of publishers' norms established by standardized tests.

# 7.4 Notice to Unit Members

No later than October 1 of the year in which the evaluation is to take place, the District shall give unit members to be evaluated a copy of the evaluation procedures, the criteria upon which the evaluation is to be based, the governing board's established standards of expected pupil achievement at each grade level in each area of study, and the identity of their evaluator. In addition, the District shall give each probationary and temporary unit member the name of an experienced person who will provide assistance to the unit member.

# 7.5 **Pre-Conference and Evaluation Plan**

No later than October 15, the unit member being evaluated and the evaluator shall meet to establish:

- 7.5.1 Standards to be achieved in the areas described in section 7.2 during the evaluation period and a complete description which may deter the achievement of such standards; and
- 7.5.2 An evaluation plan which shall include the focus of each observation and a preliminary schedule of dates for formal classroom observations, post-observation conferences and the final summary evaluation. For temporary and probationary unit members, the first formal observation shall be conducted no later than November 1; and
- 7.5.3 By mutual agreement, the evaluator and evaluatee may include any of the following as documentation of progress toward meeting District standards in the areas of evaluation described in section 7.2: video tapes, curriculum units, teacher journals, logs and calendars, interviews, lesson plans, evidence of communications with parents/students, examples of student work, records of participation in school improvement efforts, and reports on professional growth activities.

# 7.6 Classroom Observation/Post Observation Conference

Each evaluation plan shall include a minimum of two (2) formal classroom observations. Each formal classroom observation shall be a minimum of thirty (30) minutes. Prior to the formal observations, the District shall give the unit member a minimum of two (2) days' notice. Within five (5) days after each formal observation, the evaluator shall schedule a Post Observation Conference. In addition to the formal observation, the evaluator shall informally observe the unit member at least twice during each evaluation year.

With agreement of both the unit member and the evaluator, the number of formal and informal observations may be reduced to three (3) or two (2) observations.

#### 7.7 Formal Evaluation Summary

#### 7.7.1 Dates for Completion

All final evaluation conferences shall be completed by May 1 except for probationary and temporary teachers for whom the final evaluation report shall be completed on or before February 15<sup>th</sup>. No later than five (5) duty days after the final evaluation conference, the evaluator shall prepare a Formal Evaluation Summary. The evaluator shall transmit a copy to the unit member and the original to the Human Resources Office. All copies must be signed by the evaluator and the evaluatee.

# 7.7.2 Addendum to the Formal Evaluation Summary

If the unit member is given a rating of "Progress Towards Standards Not Evident" on any category of the Formal Evaluation Summary, the District must include the following on the Addendum to the Formal Evaluation Summary:

- (a) a list of areas performed in an unsatisfactory manner below District standards;
- (b) a description of improvement required, and programs/people available to provide assistance;
- (c) a time period for demonstrating improvement; and
- (d) a description of methods used to measure improvement.

The unit member is responsible for satisfactory performance. Merely completing improvement activities may not demonstrate target behavior.

7.7.3 <u>Unsatisfactory Evaluation</u>

If a unit member is rated "unsatisfactory" on a Formal Evaluation Summary, the evaluator shall hold a conference with the unit member prior to the issuance of the Formal Evaluation Summary. The unit member shall be notified of the right to have an Association representative present at the meeting. Dismissal and/or suspension of permanent unit members shall follow the standards and procedures contained in Education Code Sections 44932 through 44945.

A unit member with permanent status must participate in the Peer Assistance and Review Program if he or she receives a Summary Rating of "unsatisfactory" on the Formal Evaluation Summary.

# 7.7.4 <u>Response to Evaluation</u>

The evaluatee shall have the right to initiate a written reaction or response to the evaluation and that response shall become a permanent attachment to the evaluatee's personnel file.

# 7.8 **Personnel Files Relating to Evaluations**

- 7.8.1 Materials in unit members' personnel files are to be made available for the inspection of the unit member involved, except items which are excluded by the Education Code.
- 7.8.2 Every unit member shall have the right to inspect such materials upon request, provided that the request is made at a time when the unit member is not actually required to render services to the District.
- 7.8.3 Information of a derogatory nature, except as provided by law, shall not be entered or filed unless and until the unit member is given notice and an opportunity to review and comment on the information. A unit member shall have the right to enter, and have attached to any such derogatory statement, the unit member's own comments on the information. The unit member's review of the information shall take place

during normal business hours, and the unit member shall be released from duty for this purpose without salary reduction.

# 7.9 Complaint Procedure

No negative and unsatisfactory evaluations shall be based upon information or material of a derogatory or critical nature which has been received by the evaluator from students, parents, and/or citizens unless the following procedures have been followed:

- 7.9.1 If the complaint by a student, parent or citizen about a unit member may be used against the unit member, it shall be reported to the unit member within five (5) working days of the receipt of the complaint.
- 7.9.2 The administration shall inform the unit member of his/her right to have Association/legal representation at any meeting regarding a complaint.
- 7.9.3 Should the involved unit member believe the allegations in the complaint warrant a meeting, the immediate supervisor shall attempt to schedule a meeting between the unit member and the complainant.
- 7.9.4 If the matter is not resolved to the satisfaction of the complainant, he/she shall put the complaint in writing and submit the original to the unit member's supervisor. The supervisor shall provide a copy to the unit member. The unit member shall be given time during the duty day, without salary deduction, to review the complaint. If no written complaint is received, the complaint shall be dropped.
- 7.9.5. A unit member will be granted the right to meet with the complainant(s) as to the validity and accuracy of the complaint. This section is not intended to supersede any laws protecting confidentiality in reporting incidents falling under the category of sexual harassment.
- 7.9.6 The unit member is provided the right of representation and is informed prior to any meeting which may involve criminal allegations, that the member is entitled to such representation.
- 7.9.7 The unit member has the right to attach written and signed comments to any written complaint filed by the complainant.

- 7.9.8 If the matter is not resolved and the unit member believes the complaint is false and/or based on hearsay, or that improper procedures were followed, a grievance may be initiated.
- 7.9.9 False/Unsubstantiated Complaints

Complaints which are shown to be false, or are not sustained by the grievance procedure shall neither be placed in the unit member's personnel file nor utilized in any evaluation or disciplinary action against the unit member.

# 7.9.10 Discrimination Complaints

The complaint procedure set out in section 7.9 shall not apply to the investigation of complaints filed under Federal and/or State law. Complaints Not Subject to This Procedure: Complaints alleging discrimination, harassment and/or suspected child abuse by a unit member shall not be subject to this Complaint Procedure.

# 7.10 Status in Position

The unit member is allowed to remain in the position until formal charges have been made by the complainant, except in circumstances where the Superintendent may remove the unit member from duty with pay pending investigation of any formal complaint as permitted by Section 16.5.

## 7.11 Alternative Professional Growth Evaluation Plans

- 7.11.1 With mutual consent between a permanent unit member and the unit member's evaluator, the unit member may participate in the Alternative Professional Growth Plan. If an evaluator is unable to agree to the unit member's participation in the alternative professional growth evaluation plan, the evaluator shall complete and sign the "Selection of Evaluation Plan" portion of the Teacher Work Plan required by Section 7.5.
- 7.11.2 Each participating permanent unit member shall develop an action research question and collect evidence through a portfolio to address the research question. Each permanent unit member's plan shall be based on teacher investigation, data

collection, analysis and/or research. Each permanent unit member's plan shall include the related areas of evaluation listed in Section 7.2.

- 7.11.3 Each participating permanent unit member and site administrator shall comply with the timeline established by Article 7 and the dates specified in Section 7.4, Section 7.5, and Subsection 7.7.1.
- 7.11.4 The Alternative Professional Growth Evaluation Plan is attached to the collective bargaining agreement as Appendix D.

#### **ARTICLE 8 - ASSIGNMENTS, REASSIGNMENTS AND TRANSFERS**

# 8.1 Assignments and Change of Assignments

- 8.1.1 For purposes of this Article, "assignment" is the designation of a specific position or responsibilities within a school or department or work location. Assignment includes the initial placement of a newly employed bargaining unit member in a specific school or work location as well as the change of a unit member's position or responsibilities within the same school, department or work location. "Assignment" does not include any extra-curricular duty.
- 8.1.2 Principals will assign bargaining unit members based on the unit member's credentials, teaching experience in a subject matter or grade level, training, major and minor fields of study, advanced degrees related to the assignment, documented strengths or weaknesses, language needs of the school community, seniority (including the need for experienced staff at higher needs schools), and documented staffing needs for the effective operation of the school. The District shall not change a member's assignment for arbitrary, capricious or discriminatory reasons.
- 8.1.3 For purposes of this Article, "change in assignment" is any change in a unit member's "assignment" as defined in Section 8.1.1.
- 8.1.4 Principals shall solicit information from staff regarding assignment requests for the following school year. The principal will consider this input when making staff assignments.
- 8.1.5 The principal shall confer with the unit member to discuss any proposed change of assignment. Written notice of a change of assignment shall be given to a unit member during a conference with the originator of said change prior to the change of assignment. The unit member may request that the specific reasons for the change be given to them in writing.
- 8.1.6 A list of tentative staffing assignments for the next school year shall be posted in at least two (2) conspicuous locations in each school by May 15. The list will also be

sent to the CETA office. Unit members will be notified in writing as soon as possible of any change of assignment made after the close of school.

8.1.7 A change in assignment shall not be made arbitrarily or capriciously.

# 8.2 Reassignment Due to Layoff

- 8.2.1 For purposes of this Article, "reassignment" shall be defined as a change to a different program and/or a movement from one work site to another as a result of layoff pursuant to Education Code Section 44955, and which may be necessary to assure the retention of certificated unit members with seniority greater than those laid off. This definition specifically excludes any transfer or change in program assignment which does not result from the District's obligation to reassign pursuant to Section 44955 of the Education Code.
- 8.2.2 Any unit member scheduled to be reassigned shall be entitled to a written notice of such reassignment within ten (10) calendar days of the effective date of such reassignment and, in addition, upon request, will be entitled to a conference with the principal and the Superintendent regarding the reassignment, at which time the unit member will be provided with the reasons for the reassignment and with the opportunity to discuss those reasons.

## 8.3 Unit Member-Initiated (Voluntary) Transfers

- 8.3.1 For purposes of this Article, a "transfer" shall consist of a change in work location of a unit member from one school or work site to another school or work site within the District. Such a transfer does not encompass the process of assignment to a specific position and responsibilities within the school, or department, or work location. A unit member assigned to more than one work site shall be considered as being transferred only when moved from one District-wide program to another program.
- 8.3.2 A "vacancy" is any new position, an opening arising from a resignation, retirement, or termination, any position to which a unit member is not assigned or which is not committed for purposes of leaves, unresolved involuntary transfers or layoffs.
- 8.3.3 From August 1 through October 1 of each year, the District retains the right to adjust and add positions without posting them if the openings are due to unforeseen circumstances.
- 8.3.4 When a unit vacancy occurs after October 2 for the current school year, in order to limit disruption, the District may fill the position with an external applicant for the duration of the school year. The position, however, will be listed as an opening for the following year pursuant to Section 8.3.6 below.
- 8.3.5 In order to identify potential opening for the upcoming school year, the District may, at its discretion offer an incentive for early notification of retirement or resignation for any member at step 10 or higher. The incentive will be publicized to members by December 1. The member must submit the retirement or resignation on a Transfer/Change of Status Request form to Human Resources by January 15 to receive the incentive.

8.3.5.1 A unit member who has achieved Progress Evident or above on the most recent evaluation may, on the Transfer / Change of Status Request form, designate his/her request to leave the current site. If the request is made for three consecutive years, and the member did not refuse an offer to move during the period, a transfer to another site shall be granted. The transfer destination will be determined by the District with consideration given to employee preference.

8.3.6 On or before January 22, a posting will go out internally for all known vacancies for the next school year. Interested applicants shall apply through the District's internal application process. Depending on the number of applicants, there may be an interview process for the position. Internal interviews, if held, will be scheduled by February 28. In the event candidates for a position are substantially equal based on the criteria specified in section 8.1.2, the candidate with the greater seniority shall be recommended. The principal shall make the final decision.

- 8.3.7 The unit member requesting transfer shall be notified in writing within ten (10) working days of the interview by the Human Resources office of the acceptance or denial of the transfer.
- 8.3.8 If a voluntary transfer is denied and the unit member requests reasons for the denial, the appropriate site administrator shall meet with the unit member to discuss the specific reasons.
- 8.3.9 All posting on or after March 1 will be concurrently posted internally and externally.
- 8.3.10 On and after March 1, notices of vacancies shall remain posted for at least five (5) working days. Vacancy notices shall be posted on EdJoin and/or other platforms and sent through the District electronic mail system as soon as the District determines that a vacancy exists. The posting shall include the position description and location, grade level or subject matter assignment, credential and other special requirements.
- 8.3.11 Interviews after March 1 shall be conducted by a committee. Each Interview Committee shall include the site administrator or designee and a CETA representative. By June 1 annually the District and CETA will create a list of available CETA representatives from all appropriate grade level groupings/departments to participate in interviews for the following twelve (12) months. Additional Committee members shall be selected by the school staff.
- 8.3.12 The Committee shall submit a recommendation in writing to the principal. In the event candidates for a position are substantially equal based on the criteria specified in Section 8.1.2 the candidate with the greater seniority shall be recommended. The principal shall make the final decision.

#### 8.4 Involuntary Transfer Procedures

8.4.1 The District shall seek volunteers before involuntarily transferring a unit member to fill a vacancy unless such unit member is being transferred to improve his/her performance pursuant to Section 8.4.2. Unit members to be involuntarily transferred, other than those being transferred to improve performance pursuant to Section 8.4.2, shall have the right to indicate preferences in writing from a list of known vacancies.

The Superintendent or designee shall honor such preferences, unless extenuating circumstances exist, in accordance with the criteria in Section 8.1.2.

- 8.4.2 An involuntary transfer may be initiated by the District due to: changes in enrollment; school closure and openings; staffing shortages or surpluses within a school and/or department; verified curricular needs; to improve performance as documented by evaluation(s). In a non-evaluation year other substantive documentation may be used to justify an involuntary transfer to improve performance. No unit member shall be involuntarily transferred for arbitrary, capricious or discriminatory reasons.
- 8.4.3 Written notice of an involuntary transfer, and specific reasons when requested by the unit member, shall be given to the unit member during a conference with the originator of said change prior to a final involuntary transfer. Five (5) day notice shall be given before the transfer takes effect.
- 8.4.4 Unit member(s) involuntarily transferred will be placed in open positions comparable as possible to the previously held position(s).
- 8.4.5 No unit member shall be involuntarily transferred to create a vacancy for a unit member who will be transferred to improve performance.
- 8.4.6 Any unit member who is involuntarily transferred shall not be involuntarily transferred for two (2) school years. If the involuntarily transfer is due to enrollment changes, the unit member(s) will have the first rights of return.

## 8.5 Relocation Assistance

- 8.5.1 When the District involuntarily transfers or involuntarily changes the assignment of a unit member, the District will provide packing materials for the unit member's materials. In addition, the District will transport the boxes and equipment.
- 8.5.2 Unit members involuntarily transferred or involuntarily required to change assignments during the school year shall be given up to three days of preparation time to prepare for the new duties.

#### **ARTICLE 9 - HOURS AND DAYS OF EMPLOYMENT**

## 9.1 Professional Day

Unit members shall work a professional day. Unit members shall arrive at the school site at least twelve (12) minutes before the student instructional day and shall be in their classrooms with the door open at least three (3) minutes before the student instructional day begins. A unit member's professional day shall end upon the completion of the unit member's duties unless a "mandatory" meeting (faculty meeting, collaboration time, or training) described in Sections 9.4 and 9.9 has been scheduled.

- 9.1.1 Collaboration time is provided to allow for a systematic process, in which people work together, interdependently, to analyze and impact professional practice in order to improve student learning.
- 9.1.2 Legally Mandated Trainings shall be completed during the member's professional day and outside of "mandatory" meetings.

### 9.2 Instructional Minutes

Unit members assigned to TK-3 classroom instruction shall instruct students in the classroom for no less than 50,220 minutes and no more than 50,520 minutes per school year. Unit members assigned to classroom instruction beyond third grade at elementary sites shall instruct students in the classroom for no less than 49,918 minutes and no more than 50,218 minutes per school year. Unit members assigned to classroom instruction at a middle school shall instruct students in the classroom for no less than 46,620 minutes and no more than 46,920 minutes per school year. To meet program needs and provide program flexibility, individual school sites may exceed the minutes specified in this section by complying with the waiver procedures established in Article 18.

### 9.3 Adjunct Duties

Unit members at each site shall elect at least three (3) unit members to act as the site's Building Advisory Committee. After receiving input from the site administration and the site members, each site's Building Advisory Committee shall develop a list of adjunct duties, and unit members shall self-select adjunct duties. Unit members employed on partial contracts shall perform adjunct duties on the same basis as their employment bears to full-time employment. If necessary, the Building Advisory Committee shall make final assignments.

#### **MOU for 2023-2025:**

The representatives of Campbell Union School District (District) and the Campbell Elementary Teachers Association (CETA), collectively the "Parties" agree to and enter into this Memorandum of Understanding (MOU), for two years effective July 1, 2023 - June 30, 2025. The purpose of this MOU is to pilot the impact and effectiveness of adding District/site initiatives to the adjunct duty list.

- 1. Each site's BAC list of adjunct duties will include up to two District/site initiatives selected by the site administrator.
- 2. The site administrator shall provide the list of District/site initiatives to the Site Representatives within the first week of the school year.
- 3. The BAC will ensure the District/site initiatives are filled.
- 4. The District and the Association will meet by April 15, 2025 to assess the effectiveness of this pilot.

The Parties will make every effort to come to mutual agreement on the issue of adjunct duties as part of the 2025 Successor negotiations prior to June 30, 2025. If the parties have not resolved the issue by June 30, 2025, the terms of the pilot MOU will apply to the 2025-2026 BAC adjunct duty process and this agreement will expire without precedent on June 30, 2026.

### 9.4 Mandatory Meetings

Mandatory meetings as described in 9.1 shall be scheduled for an average of one meeting per week. At the elementary sites, the weekly meeting will generally consist of the following: collaboration, staff meetings, and one prep of at least 50 minutes per month. This prep will be in addition to the prep described in 9.8.1. Normally, meetings will be scheduled on Wednesdays. Teachers in middle school may have ten (10) meetings per year beyond the average of one meeting per week. Meetings shall last no more than ninety (90) minutes. In the event that a mandatory meeting requires more than ninety (90) minutes, the principal reserves the right to extend up to four (4) such meetings to no longer than 120 minutes. An attempt will be made by the administration to notify staff of such an extension as soon as

possible but no later than the Friday prior to the meeting. Each site will develop a meeting protocol to maximize the effectiveness and efficiency of the meetings. Generally, meetings should start approximately fifteen (15) minutes after the last student dismissal.

#### 9.5 Coordination with CETA Meetings

In May of each school year, CETA shall give CETA's Monday meeting schedule through December of the next school year to the Assistant Superintendent – Human Resources. The District shall not schedule mandatory meetings after 3:30 p.m. on scheduled Monday CETA Rep Council meeting days. If the District schedules a Monday meeting that conflicts with a CETA Executive Board or committee meeting, impacted CETA members shall be excused to attend the CETA meeting.

## 9.6 Night Meetings

Unit members shall attend one "Back to School" night per school year. Unit members shall also meaningfully participate in one after-school hours event, scheduled between January and June. The site Leadership Team (LT) shall determine the list of events no later than December 31. The events chosen will focus on celebrating student learning and growth. CETA may add a representative for consideration of this issue only. The site LT may add or change the list of eligible events as it deems appropriate to meet the above stated needs. The event(s) will be aligned to the school's specific LCAP goals or objectives and will have a design and schedule to foster meaningful parent engagement.

# 9.7 Duty-Free Lunch

Unit members shall have a daily, thirty (30) minute, duty-free lunch period. If the students at the school site have a lunch break longer than thirty (30) minutes, unit members at that site shall have a lunch break as long as the students' lunch break. In the event of inclement weather, unit members' duty-free lunch period may be shorter than the student lunch break, so that unit members can provide indoor classroom supervision.

# 9.8 Unit Member Preparation Time

The purpose of teacher directed preparation time is to provide classroom teachers the opportunity to plan and conduct activities related to the delivery of instruction and manage the many requirements of the job. Generally, teachers are expected to be on campus during their preparation time. If there is a need for a teacher to leave campus for an activity relating

to the delivery of their instruction, the teacher will notify the administrator. Teachers are required to sign out in the school office if leaving campus during their work day.

9.8.1 <u>Middle School</u>

The middle school classroom unit members shall be provided with a daily preparation/consultation period equal to a normal instructional period at that site. For middle schools on Flexible Schedules, the weekly preparation time shall be equivalent to the weekly amount of time.

# 9.8.2 Elementary School – including TK-8 Schools – Upper Grades

Unit members teaching above grade 3 in elementary school classrooms shall receive one hundred and eight (108) preparation periods per year during the student instructional day. A preparation period shall equal fifty (50) minutes. Such preps will be scheduled routinely throughout the year. In addition, the District shall provide upper grade unit members at elementary schools with one release day during the school year. The release day shall be used by unit members for instructional planning and preparation. Preparation time for teachers in grades 3 and 4 combination classes that have 24 students or less shall be the same as the TK-3 teachers. Preparation time for teachers in grades 3 and 4 combination classes that have more than 24 students shall be the same as for upper grades.

9.8.3 <u>Grades TK-3</u>

For the 2023-24 school year, grades TK-3 classroom unit members shall receive thirty-six (36) preparation periods per year during the student instructional day. A preparation period at the grades TK-3 level shall equal fifty (50) minutes. Such preps will be scheduled routinely throughout the year. Additionally, for the 2023-24 school year only, TK-3 teachers will be released for one half-day for teacher directed prep.

Effective July 1, 2024, grades TK-3 classroom unit members shall receive forty-five (45) preparation periods per year during the student instructional day. A preparation period at the grades TK-3 level shall equal fifty (50) minutes. Such preps will be scheduled routinely throughout the year

### 9.8.4 <u>Preparation Time Teachers</u>

Preparation time teachers shall not be entitled to preparation periods.

# 9.8.5 <u>Elective Teachers</u>

Elective teachers in elementary and TK-8 schools shall receive the amount of preparation time that corresponds to the grade range that makes up the majority of their teaching schedule. Elective teachers shall receive the higher amount of prep time in cases where that majority overlaps multiple grade ranges.

# 9.8.6 <u>Substitutes for Preparation Time Teachers</u>

If an elementary preparation period teacher is absent, the District will employ a substitute as needed to ensure the annual allotments noted in 9.8.2. If no substitute is available, the procedures established in 10.10.1 shall be followed.

## 9.8.7 <u>SDC Pre-School Teachers</u>

Special Education Pre-School Teachers shall receive one fifty (50) minute preparation period per week. The schedule will be determined between staff and the Director of Special Education. Such preparation time shall be scheduled so as to 1) maintain the integrity of the special education program for special education students, and 2) be at no additional cost to the district.

# 9.8.8 <u>Missed Preps</u>

In the event that a unit member's preparation period at an elementary or K-8 school must be canceled for reasons beyond our control, the District shall make reasonable attempts to reschedule the missed prep. In the event that the prep cannot be rescheduled, as a last resort, unit members shall be compensated for the missed prep at one and one-half  $(1 \frac{1}{2})$  times the hourly rate in their June paycheck.

# 9.9 Shortened/Minimum Student Instructional Days

The last two (2) days of the school year shall be scheduled as shortened days. Wednesdays shall be designated as defined in 9.4.

## 9.9.1 <u>Use of Shortened Days</u>

The District shall schedule the following shortened days:

Five (5) shortened instructional days each school year for parent-teacher conferences in the fall. Student release during the fall conference week will be approximately an hour earlier than the typical minimum day at a site to allow for additional time to conference with parents. Five (5) shortened instructional days each school year for parent-teacher conferences in the fall.

At the middle school, an additional three (3) shortened Wednesdays shall be designated for parent-teacher communication.

## 9.10 Parent/Teacher Intervention Conferences

Parent/Teacher Intervention Conferences shall be held for students who are at risk of retention as is required by law, school district policies and regulations. Two (2) weeks in the second trimester will be designated by the district as weeks where no mandatory meeting will be held.

# 9.11 Length of Work Year

The work year for unit members assigned to the classroom shall be one hundred eighty-five (185) days. The placement of unit members' duty days and the work calendar shall be negotiated and attached as Appendix B to this Agreement.

## 9.12 Election Day Meetings

On local, state, and/or national election days, no mandatory school or District meetings shall be scheduled.

#### **ARTICLE 10 - COMPENSATION**

Compensation for 2023-2024, effective July 1, 2023, shall a 7% increase on the salary schedule.

#### **10.1** Salary Schedule

- 10.1.1 Unit members shall be compensated on the salary schedule set forth in Appendix A and incorporated into this Agreement. Compensation and stipends are based on 1.0 FTE and will be pro-rated for part-time.
- 10.1.2 <u>Salary Schedule Regulations</u>
  - 10.1.2.1 Advanced Degree Payment: Annually, \$2,800 shall be paid to unit members holding a Master's or Education Specialist (EdS) degree. Salary credit will be given for one (1) advanced degree only. Payment for a advanced degree received during the school year will be prorated. The annual advanced degree payment shall be incorporated into the unit member's base salary.

## 10.1.2.2 <u>Supermaximum Increments</u>

(1)Supermaximum increments of \$2,500 per annum will be paid at experience years fifteen (15) Column III.

- (2) Supermaximum increments of \$2,250 per annum will be paid at experience year twenty (20) in Column III.
- (3) Supermaximum increments of \$2,500 per annum will be paid at experience year twenty-four (24) in Column III.
- (4) Supermaximum increments of \$2500 per annum will be paid at the experience year twenty-five (25) column III.
- (5) Prior experience recognized by the District upon hiring will be counted toward supermaximum pay scale, except military service performed prior to becoming a District employee.
- (6) Years of service in the Campbell Union School District must be consecutive as recognized by the current salary schedule policy.

- (7) Supermaximum increments shall be incorporated into the unit member's base pay and shall be prorated for less than full time unit members.
- (8) The revisions to Section 10.1.2.2 and the change in years of experience for supermaximum increments was negotiated pursuant to Government Code Section 3543.2(d). The Association agrees to waive any right to a claim under Education Code Section 45028, and further agrees not to support any claim, grievance, or complaint based upon an alleged violation of Education Code Section 45028.
- (9) Unit members referred to PAR, but not self-referred, shall not earn step or column movement during the period they are in PAR. The PAR Joint Panel shall ensure that the appropriate referral process is followed.

#### 10.1.2.3 BCLAD Stipend

The District shall pay a two thousand dollars (\$2,000) annual stipend to a unit member possessing the BCLAD credential. The BCLAD stipend shall be incorporated into the unit member's base pay.

# 10.1.3 Salary Credit for Training

- 10.1.3.1 In Column B I and above, unit members must hold a valid credential to cover their teaching assignment.
- 10.1.3.2 Units for Column I, II, and III must be from courses whose content is relevant to the subject/grade level taught and/or will better prepare the unit member to further the District toward its educational goals.
- 10.1.3.3 <u>Continuing Education Units (CEUs)</u>
  - (1) Salary credit for CEUs will be granted with the following considerations:

- (a) CEUs must receive approval for salary credit from the Human Resources Department prior to enrollment in the course.
- (b) CEUs receiving approval must be relevant to the subject/grade level taught and/or better prepare the unit member to further the District toward its educational goals.
- (c) Credit will be granted for CEUs at a rate of 2/3 (.666) of a semester unit.
- 10.1.3.4 Any units (semester, quarter or CEUs) for which salary credit is granted must:
  - Be taken from an accredited university, college or pre-approved CEU institution;
  - (2) Have received Human Resources Department approval prior to enrollment in the course;
  - (3) Have been taken subsequent to the completion of a Baccalaureate Degree;
  - (4) Be verified by an official transcript (tentative verification can be submitted by means of a letter of completion or grade card); and
  - (5) Be submitted to the Human Resources Department by August 31st of the year in which salary credit is to be granted.
- 10.1.3.5 Salary credit will be granted once per year effective beginning on the September pay warrant for continuing unit members or on the date of hire for new unit members.
- 10.1.3.6 Salary Credit Appeals Committee
  - The Salary Credit Appeals Committee will be made up of six (6) members including three (3) unit members representing K-4, 5-8 and Special Education, and three (3) administrators.

- (2) The Salary Credit Appeals Committee will meet once per year, prior to May 15th to hear and rule on appeals. The Salary Credit Appeals Committee's decision is final.
- (3) A unit member who has had units rejected by the Human Resources Department may appeal that ruling to the Salary Credit Appeals Committee in writing prior to May 1st of the year the appeal is to be heard.
- 10.1.3.7 A list of approved areas of emphasis for course work will be posted annually at each work site. This list will be developed by the Instructional Department.
- 10.1.4 Credit for Teaching Experience
  - 10.1.4.1 Credit for teaching experience outside the Campbell Union School District will be granted as follows:
    - The teaching must have been in an institution that required all of the teachers in the school to be credentialed.
    - (2) The teachers in the institution must have received a four (4) year college degree and a regular teaching credential prior to rendering service.
    - (3) Effective July 1, 2016, in instances of hard to staff areas, the District may give salary placement credit for training and nonteaching experience that is directly related to the teaching assignment.
    - (4) Effective July 1, 2016, in instances in which the member is transferring from a state or program in which the credential was embedded into a four-year program, those members will be deemed to have 30 units and to have met the requirements for placement on Column I of the salary schedule. All additional units acquired after hire will be applied towards column advancement.

- 10.1.4.2 Full credit for teaching experience in the Campbell Union School District will be granted if the unit member returns to teaching after an absence of less than ten (10) years. If a teacher returns after an absence of ten (10) consecutive years or more, experience credit will be granted at the Superintendent's discretion.
- 10.1.4.3 Credit for Nursing Experience

Certificated employees hired by the District on or after July 1, 2023, to serve as a school nurse shall receive year-for-year credit for all employment as a registered nurse up to a maximum of fifteen (15) years.

- 10.1.4.4 Credit for active military service will be allowed in lieu of out-of-District teaching experience to a maximum of five (5) years.
- 10.1.4.5 Certificated employees hired by the District on or after July 1, 2023, credit for teaching experience outside the District will be granted for a maximum of fifteen (15) years.
- 10.1.4.6 This provision was negotiated pursuant to Government Code Section 3543.2(d). The Association agrees to waive any right to a claim under Education Code Section 45028, and further agrees not to support any claim, grievance, or complaint based upon an alleged violation of Education Code Section 45028.

# **10.2** Salary of Employees Not Completing Year of Employment

10.2.1 Any unit member who leaves the employ of the District before the end of the school year shall be paid in proportion to the number of days he/she served.

### **10.3** Salary Warrants

10.3.1 All full-time personnel will be paid on the last working day of each month.

## **10.4** Repair or Replacement of Personal Property

10.4.1 It shall be the Campbell Union School District's policy to repair or replace personal property of District employees when such property is damaged or stolen while the

employee is in the line of duty, without fault of the employee, in accordance with administrative rules and regulations.

10.4.2 With the exception of vehicular damage/loss, the District shall not make any compensation in excess of six hundred dollars (\$600) per employee per incident.

## **10.5** Health and Welfare Benefits

Health and Welfare Benefits include Medical, Vision and Dental

- 10.5.1 Beginning July 1, 2023, the District shall contribute the cost of Kaiser HMO Employee Only, Delta Dental PPO, and VSP Vision for each full-time bargaining unit member, or up to <u>thirteen thousand eight hundred thirty-seven dollars (\$13,837)</u>, whichever is less.
- 10.5.2 For part-time unit members, the District shall pay a fractional amount of the individual maximum amounts as determined by the portion that is equal to the fraction of hours worked to full-time employment, only if the part-time unit member contributes by payroll deduction an amount equal to the balance.
- 10.5.3 Durable medical equipment shall be added to the plan so long as the medical plan offers this benefit.
- 10.5.4 This section shall apply to domestic partners who show evidence they are registered with the Secretary of State.
- 10.5.5 <u>Married and Domestic Partners, Both of Whom are CUSD Employees</u> The District's contribution for two employees who are married to each other or who are domestic partners under the legal definition shall not exceed a total of twenty-six thousand eight hundred sixty-eight (\$26,868) per year, payable in monthly amounts. Beginning July 1, 2021, the District shall contribute the cost of Kaiser HMO Employee Only, Delta Dental PPO, and VSP Vision, per employee, or twenty-six thousand eight hundred sixty-eight (\$26,868), whichever is less. The amount shall be prorated for part-time unit members.

43

## 10.5.6 <u>Health Insurance Pool</u>

The District shall maintain a Health Insurance pool intended to be used to offset the out-of-pocket costs incurred by eligible members who utilize two-party or family coverage. The pool shall be funded annually with a "base amount", which for the payout due in December of the 2011-2012 school year will be \$63,426. In future years, the "base amount" shall be increased by the same percentage as the increase to the medical health and welfare cap, if any.

When an increase to the health benefit contribution occurs, the base amount of the benefit pool will be increased by the equivalent percentage of the increase to the health benefit cap. The new base amount will be divided among the eligible members' FTE. In order to be eligible for the benefit pool, the member must have been employed and paying out of pocket for a two-party or family health benefit plan for the entire school year. Eligible members will receive their portion of the benefit pool payout in their last pay warrant of the school year.

# **10.6 Extended Duty Pay**

10.6.1 The hourly rate for Extended Non-Instructional Duties and Student Extended Day Instructional or Intervention Programs shall be \$50 per hour.

#### **10.7** Pay for NBPTS Certification

## 10.7.1 District Stipend

Effective July 1, 2000, unit members who achieve certification through the National Board of Professional Teaching Standards will receive an annual stipend of \$2,000. This annual stipend shall be incorporated into the base pay, and will be prorated for unit members employed less than full time.

### 10.7.2 Start Date

This stipend will be paid beginning with the school year following notification to the unit member that NBPTS certification has been achieved, and will continue for each year for which NBPTS certification is valid.

#### 10.7.3 State of California Stipend

If the unit member receives a State of California stipend greater than the District stipend defined in this section, the District stipend shall not be paid to the unit member in the same school year in which the unit member receives the State of California stipend.

#### 10.8 Stipends

10.8.1 Stipends for After School Sports

Unit members may apply for and be selected for assignments in the after school sports program. A unit member's assignment in the after school sports program shall be in addition to the unit member's full-time assignment and may be terminated by the District at any time. Whenever possible, the District shall offer bargaining unit members vacant positions in the after school sports program before employing non-bargaining unit walk-on coaches. Unit members shall be compensated on this After School Sports Stipend Schedule:

Assignment	Stipend
Coach	\$900 per sport
Athletic Director	\$7,000 per year

- 10.8.2 Stipends for Speech and Language Specialists, Occupational Therapists and Special Education Unit Members
  - A stipend of \$2,000 will be paid in equal monthly installments as a recruitment incentive during the first year of service to CUSD, provided the unit member has at least a preliminary credential.
  - An additional \$2,000 retention incentive will be paid in equal monthly installments annually during each year of service to CUSD, provided the unit member has at least a preliminary credential.
  - The district may extend the work year for speech and language specialists, not to exceed five additional days. For each day that may be extended, the speech and language specialist will earn their per diem rate of pay.

- 10.8.2.1 Effective July 1, 2019, the provisions stated in 10.8 shall also apply to Occupational Therapists.
- 10.8.3 Middle School Activities Directors shall receive \$3,000 annually, paid in equal monthly installments.

# 10.9 Teacher Travel and Overnight Supervision

- 10.9.1 Unit members who are required and approved to go from one school to another in the course of their duty and unit members who use their car(s) for field trips and to attend special required evening meetings shall receive a mileage allowance allowed by the Internal Revenue Service. Costs to and from home to regular work assignments are not included.
- 10.9.2 The maximum round trip distance allowed for reimbursement for unit members returning to the District from their home on any given workday for required meetings shall be twenty-five (25) miles.
- 10.9.3 Teachers shall receive an additional stipend of one hundred dollars (\$150) per night for overnight student supervision for any Board-approved overnight activity.

### **10.10** Substitute for Absent Teacher

10.10.1 Elementary School

In the event that an elementary classroom that is supposed to be supervised by a substitute is left without supervision, the District agrees to make best efforts to secure alternative coverage, which may include non-instructional credentialed staff, prior to splitting the students among other teachers' classrooms.

10.10.1.1 When the best option for making certain students are appropriately supervised is to move the students into other classrooms, the impacted teachers shall be compensated after the first hour as follows: 1-3 hours = \$95 split between the impacted teachers; over 3 hours and up to a full day = \$190 split between the impacted teachers. For example, if a class is split between 2 teachers for 4 hours, each teacher will receive \$95.

10.10.1.2 If an elementary teacher who has a prep during the first hour agrees to supervise a classroom during the prep for the reason described in 10.10.1 above, that teacher shall receive one hour of pay at the hourly rate.

### 10.10.2 Middle School

10.10. 2.1 Volunteering unit members shall be compensated for substitute teaching during their prep period at one and a half the rate in section 10.6.2. Unit members shall be paid monthly.

10.10.2.2 In response to having no volunteers as described in 10.10.2, each unit member in grades 6-8 may be required to substitute up to two times per year. The unit member(s) shall be compensated at one and a half the rate in section 10.6.2.

10.10.2.2.1 When unit members in grades 6-8 volunteer to substitute teach during their prep period, those occasions shall count towards the two required dates.

10.10.2.2.2 By the Friday of the first week of school, each site will post a digital alphabetical list of eligible unit members (by prep period). This dynamic list will serve to inform unit members of the order in which they will be called upon to substitute (by prep period). Unit members in grades 6-8 may request by the following Friday to be placed at the top of the list.

10.10.2.2.3 Unit members in grades 6-8 shall be allowed to postpone one required substitute date due to time pressures or deadlines in their existing workloads. Site administrators and/or designee will maintain a calendar of key site-level events (such as, 8th grade exhibitions, articulations) and will not call on impacted staff during those events. Non-tenured unit members will not be required to substitute, but may volunteer.

47

10.10.2.2.4 In the event that a 6-8 classroom is without a substitute, or when the administrator determines, the best option for making certain that students are appropriately supervised is to move the students into other classrooms, the impacted teachers shall be compensated after the first hour as follows: 1-3 hours = 95 split between the impacted teachers; over 3 hours and up to a full 4-6 hour day = 190 split between the impacted teachers. For example, if a class is split between 2 teachers for 4 hours, each teacher will receive 95.

10.10.2.3 Record keeping for this release time shall be kept by the site administrator, with a duplicate record on file at the Human Resources Office.

## **10.11** Psychologists and Nurses

10.11.1 Nurses

Bargaining unit members assigned to nursing positions shall be placed on the regular teacher salary schedule for the work day calendar described in Appendix B. Any increase in days beyond the regular work year will be paid on the per diem rate for that schedule. In addition, bargaining unit members assigned to nursing positions shall receive an annual stipend of \$2,000, effective July 1, 2021. This annual stipend shall be incorporated into the base pay, and will be prorated for unit members employed less than full time.

#### 10.11.2 Psychologists

Bargaining unit members assigned to psychologist positions shall be placed on the regular teacher salary schedule for the work day calendar described in Appendix B. An increase in days beyond the regular work year authorized in writing by the District will be paid on the per diem rate for that schedule. In addition, bargaining unit members assigned to psychologist positions shall receive an annual stipend of \$12,500, effective July 1, 2002. This annual stipend shall be incorporated into the base pay, and will be prorated for unit members employed less than full time.

# 10.12 STRS Defined Supplemental Benefits Plan

Subject to the criteria and limitations stated in Education Code Sections 22119.2, 22905, 22954 and relevant STRS regulations, the following services performed for the District shall be creditable for inclusion in the State Teachers Retirement System ("STRS") Defined Supplemental Benefit Plan.

- Supplemental Pay Sections 10.8.1 (Coaches), 10.6 (Extended Duty Pay), 10.10 (Substituting During Preparation Periods);
- Student Extended Day Instruction or Intervention Program (Section 12.2);
- Extended Non-Instructional Extra Duties (10.6.1)
- Saturday School (Section 10.6.2);
- Regular teaching within the school year beyond 1.0 F.T.E;
- Any other activity mutually agreed in writing by the Superintendent and the President of the Association as an amendment to this provision.

Both the District and the unit member shall make any contributions to STRS required by law.

#### **ARTICLE 11 - SAFETY CONDITIONS**

#### **11.1 Employee Protection**

- 11.1.1 Bargaining unit members may use such force as is reasonable under the circumstances to protect themselves from attack, to protect another person or property, to quell a disturbance threatening physical injury to others, or to obtain possession of weapons or other dangerous objects upon the person or within control of a student.
- 11.1.2 Whenever any unit member is attacked, assaulted or menaced by any student, it shall be the duty of the unit member and the site or District office administrator with knowledge of the incident to promptly report the incident to the appropriate law enforcement as required by Education Code 44014.

### **11.2 Working Conditions**

- 11.2.1 A unit member has the right to submit written recommendations to the immediate supervisor regarding the maintenance of safe working conditions, facilities and equipment repairs and modifications and other practices designed to insure compliance with applicable standards. Within ten (10) unit work days of the submission of a unit member's recommendations, the immediate supervisor shall provide a written response.
- 11.2.2 In the event a hazardous, unsafe, or unsanitary condition exists within a school making it necessary to dismiss students, unit members will not be required to remain in the building but may be reassigned to other instructional activities.
- 11.2.3 Unit members shall not be required to work under unsafe conditions or to perform tasks which endanger their health, safety, or well-being as determined by the appropriate Federal, State, or local agency. The District agrees to follow OSHA requirements concerning training of unit members and notification to unit members who, by the nature if their assignments, work with or are exposed to hazardous chemicals and/or toxic materials.

- 11.2.4 Unit members shall be provided with adequate work space and access to copy machines, student records and other equipment necessary for unit members to perform their duties.
- 11.2.5 Classrooms shall be equipped with telephones with outside lines.

## 11.3 Suspension

- 11.3.1 A unit member may suspend for the reasons listed in Education Code 48900, any student from his/her class for the day of the suspension and the day following. The unit member shall immediately report the suspension to the principal of the school and send the student to the principal for appropriate action. As soon as possible, the unit member shall ask the student's parent or guardian to attend a parent-teacher conference regarding the suspension. A school administrator shall attend the conference if the unit member or the parent or guardian so requests. The student shall not be returned to the class from which he/she was suspended, during the period of the suspension, without the concurrence of the teacher of the class and the principal.
- 11.3.2 A written description of the rights and duties of all unit members with respect to student discipline shall be distributed to each unit member.

## **11.4 Support Services**

- 11.4.1 When, in a unit member's judgment, a student's ongoing behavior interferes with instruction, he/she will inform his/her principal in writing. The principal shall meet for a conference with the unit member to discuss the problem and decide upon the steps for resolution.
- 11.4.2 When, in a unit member's judgment, a Special Education student's ongoing behavior interferes with instruction, he/she will inform his/her principal in writing. The principal shall confer with the unit member to discuss the problem and decide upon the steps for resolution based on procedures established by Special Education law.

# 11.5 Disclosure

11.5.1 The administration shall inform unit members of every student who has caused or attempted to cause serious bodily injury or injury to another person, based on any written records that the District maintains or receives from a law enforcement agency or another District regarding a student described in Section 49079 of the California Education Code. Any information received by a unit member pursuant to this provision shall be received in confidence for the limited purpose for which it was provided and shall not be further disseminated by the unit member.

## **ARTICLE 12 - SUMMER SCHOOL**

## 12.1 Employment

- 12.1.1 The District Human Resources Office shall provide summer school applications and program descriptions by March 1 of each school year.
- 12.1.2 The criteria for selecting for summer school positions shall be based on certification of the unit member, special requirements in the program description, the unit member's teaching experience in a subject matter or grade level, major/minor field of study, documented strengths or weaknesses, and affirmative action requirements.
- 12.1.3 Unit members shall be given first priority to summer school positions.
- 12.1.4 When two (2) unit members are considered equal by the District, the unit member with the most seniority shall be selected.
- 12.1.5 Except for emergencies, applicants shall be informed of their basic assignment by May 15. Basic assignments shall be subject to adequate enrollment at the beginning of summer school.

## 12.2 Compensation

Summer school unit members shall be compensated at the hourly rate established in Section 10.6.2. Summer school personnel shall be employed and compensated for two (2) non-instructional days designated for preparation.

# 12.3 Sick Leave

Unit members shall be allowed one (1) day of sick leave during the summer school session. This day shall not be part of the allotted sick leave for the regular school year and shall not be cumulative from year to year.

## **ARTICLE 13 - PARTIAL CONTRACTS, JOB SHARING, AND PART-TIME HOURLY**

## **13.1** Partial Contracts

Partial contracts shall be made available to permanent certificated unit members provided the following conditions are met:

- 13.1.1 Contracts are made by the initiation of the individual unit member.
- 13.1.2 There is no increase in cost to the District for salary or benefits beyond the cost that would be incurred for a single unit member.
- 13.1.3 Contracts are found to be legal under all aspects of tenure, dismissal and other related laws.
- 13.1.4 Contracts are agreeable to the immediate supervisor and fit specific educational needs of the school and department. If denied by the immediate supervisor, objective reasons for denial will be given to the unit member (s) upon request. This action may be appealed to the Assistant Superintendent for Instructional Services.
- 13.1.5 The Governing Board gives special approval to each individual request.
- 13.1.6 Problems involved with statutory layoff reduction in programs shall not be increased by the granting of contracts.
- 13.1.7 Benefits and salaries shall be prorated for the part-time contracts in proration to the amount that part-time employment bears to full-time employment. Proration of health benefits shall be based on the defined District's maximum contribution.
- 13.1.8 Partial or shared-contract unit members working less than seventy-five percent (75%) shall accrue service credit for annual salary advancement in direct relation to their percentage of employment. No unit member shall remain more than two (2) years on a single, existing, salary schedule step. Any unit member working seventy-five percent (75%) or more of any given school year shall receive a full year's service credit for each year worked.

# 13.2 Job Sharing

13.2.1 Job sharing may be established, upon approval by the Governing Board, when two(2) unit members jointly apply for partial contracts for purposes of sharing one full

time teaching assignment. Such joint applications shall be made directly to the Superintendent or designee. The District administration may approve job shares in which the teachers provide a comprehensive plan addressing: student learning goals, effective communication strategies, parental engagement, lesson planning, attendance to meetings and professional development, as well as implementation of the district's instructional program. The District will be more amenable to job shares in areas in which there is a shortage of appropriately credentialed candidates. Preference will be given to proposals submitted by March 1. The Shared Teaching Contract Guidelines in Appendix E provide information necessary to complete the Application.

13.2.2 Applicant agrees to remain on shared employment at the District's convenience. Subject to seniority and the Education Code restrictions, it is understood by the applicant that he/she will have first opportunity for reinstatement to a full time contract when such a position is requested by the applicant, the position is available, and the applicant is qualified by credentialing and experience to perform this service.

### **13.3** Part-Time Hourly

A part-time hourly position in categorically funded projects is considered to be a unit position less than fifty percent (50%) of a comparable regular bargaining unit position.

13.3.1 A categorical position requiring more than sixteen and one-half (16.5) student contact hours per week for the duration of that particular program will be considered a partial contract position and compensation will be pursuant to Article 13 of this Agreement.

#### **ARTICLE 14 - LEAVES**

#### 14.1 Personal Illness and Injury

#### 14.1.1 Annual Earned Sick Leave

Full-time unit members shall be entitled to ten (10) days leave with full pay for each school year for purposes of personal illness or injury. Unit members who work less than full-time shall be entitled to that portion of the ten (10) days leave as the number of hours per week of scheduled duty relates to the number of hours for a full-time unit member in a comparable position.

#### 14.1.2 Accumulated Earned Sick Leave

If a unit member does not utilize the ten (10) days of leave as authorized in Section 14.1.1, the amount not utilized shall be accumulated from year to year.

## 14.1.3 Extended Sick Leave with Difference Pay

When a unit member has exhausted all available sick leave, including the leave described in Sections 14.1.1 and 14.1.2, and continues to be absent on account of illness or accident for an additional period of up to five school months, whether or not the absence arises out of or in the course of the employment of the unit member, the amount deducted from the salary due the unit member shall not exceed the sum that is actually paid a regular substitute employed to fill the absent unit member's position or, if no substitute was employed, the amount that would have been paid to a regular substitute if one had been employed. The sick leave described in Sections 14.1.1 and 14.1.2 and the five-month period described in this section shall run consecutively. A unit member shall not be provided more than one five-month period per illness or accident. If a school year terminates before the five-month period is exhausted, however, the unit member may take the balance of the five-month period in a subsequent school year. If a district retiree is the substitute, the additional amount paid to retirees shall not be deducted under this section.

## 14.1.4 <u>Reemployment List</u>

When a unit member has exhausted all the available sick leave described in Sections 14.1.1, 14.1.2, 14.1.3, and 14.14 and continues to be absent on account of illness or accident, and is not medically able to resume the duties of the unit member's position, the unit member shall be placed on a reemployment list. Probationary unit members shall be placed on the list for a period of 24 months and permanent unit member is medically able to return during the 24 or 39 months. When the unit member is medically able to return during the 24 or 39 months, the District shall assign the unit member to a position for which the unit member is credentialed and qualified. If at the end of the 24 or 39 month period, the unit member is unable to perform the duties, the unit member shall be dismissed.

14.1.5 <u>Verification of Leave</u>

The administration may require a medical doctor's certification or proof of illness before allowing payment for days of absence due to illness, accident or quarantine.

14.1.6 <u>Return to Work</u>

A medical doctor's verification and clearance to return to work may be required for any absence of five (5) days or more for illness, accident or quarantine. If requested by the District management, a unit member shall not return to work until she/he submits a medical doctor's authorization to return to work.

- 14.1.7 Leave to Care for a Child, Parent, or Spouse
  - 14.1.7.1 In any school year unit members may use up to a maximum of six (6) days of leave, that is credited under Section 14.1, to attend to an illness of the unit member's immediate family member as defined by Section 14.3.2 of this Agreement.
  - 14.1.7.2 This section does not extend the maximum period of leave to which a unit member is entitled under the Family Medical and Leave Act of 1993 (29 U.S.C. Section 2606, et seq.), the California Family Rights Act (Government Code Section 12945.2) and District policies

implementing these Acts regardless of whether the unit member receives sick leave compensation during that leave.

## 14.2 Personal Necessity Leave

## 14.2.1 Days Per Year

Leave which is credited under Section 14.1 may be used, at the unit member's election, for purposes of personal necessity; provided that use of personal necessity leave does not exceed eight (8) days in any school year. Personal necessity leave shall be granted for a minimum of one-half (1/2) day per occurrence.

## 14.2.2 Definition

For purposes of this provision, personal necessity shall be limited to: (1) death or serious illness of a member of the unit member's immediate family; (2) an accident involving the unit member's person or property, or the person or property of a unit member's immediate family; (3) a unit member who appears in court as a litigant; or (4) other matters of compelling personal importance, provided that personal necessity leave shall not be used solely for: (a) the extension of a holiday or a vacation period; (b) matters which can be taken care of outside the work hours; (c) recreational activities; (d) participation in demonstrations, slowdowns, strike or similar work stoppages; or (e) matters of mere personal convenience.

14.2.2.1 One (1) personal necessity leave day may be used without providing a reason to the District.

## 14.2.3 Use for Wedding Events

A unit member may use no more than two (2) days of personal necessity leave to prepare for and attend the unit member's wedding ceremony. Under no circumstances may a unit member use personal necessity leave for a honeymoon or a wedding-related vacation. The unit member shall mark the "My Wedding" space on the Absence Report Form.

#### 14.2.4 Use During Staff Development

Under no circumstances may a unit member use personal necessity leave to avoid attending staff development.

## 14.2.5 <u>Requesting a Substitute</u>

Before using personal necessity leave under the circumstances outlined in Section 14.2.2, the unit member shall make every effort to comply with District procedures to enable the District to secure a substitute.

## 14.2.6 <u>Verification of Leave</u>

Under all circumstances, a unit member shall verify in writing, signed by the unit member, that the personal necessity leave was used only for purposes stipulated in this Article. A District-provided form shall be available at work locations.

#### **14.3** Bereavement Leave

- 14.3.1 A unit member shall be entitled to a maximum of three (3) days, (five [5] days if travel of 300 miles or more is required), of leave of absence without loss of salary or sick leave on account of the death of any member of his/her immediate family. This leave shall be used only for grieving and attending funeral or memorial services following a specific death. At its discretion, the District may request a copy of the death certificate or memorial/funeral program.
- 14.3.2 For purposes of this provision, an immediate family member shall be limited to: mother; mother-in-law; father; father-in-law; husband; wife; son; son-in-law; daughter; daughter-in-law; sister; sister-in-law; brother; brother-in-law; grandmother; grandfather; or grandchildren of the unit member or spouse or any relative living in the immediate household of the unit member, or anyone who, over a period of time, has held the place of an immediate family member to the unit member.

#### 14.4 Leave for Pregnancy Disability

14.4.1 Unit members are entitled to use sick leave as set forth in Section 14.1 for disabilities caused or contributed to by pregnancy, miscarriage, childbirth, and recovery

therefrom, and shall be limited to those disabilities as set forth above. The length of such disability leave, including the date on which the leave shall commence and the date on which the duties are to be resumed, shall be determined by the unit member and the unit member's physician.

14.4.2 The District may require a medical doctor's verification and clearance to return to work as provided in Section 14.1.6.

#### 14.5 Leave for Bonding with Recently Placed Foster or Adopted Child or Newborn Infant

- 14.5.1 Bonding leave will run concurrently with leave rights under the Federal Family and Medical Leave Act (FMLA) and the California Family Rights Act (CFRA). It shall also be coordinated with the pregnancy disability and child rearing leaves of this Agreement.
- 14.5.2 Regardless of the limits in 14.2.1 above, each eligible unit member may be allowed up to twelve (12) weeks; within the 12 month period immediately following the birth or placement of a child; personal necessity leave for purposes of bonding with a newborn infant, foster, or a recently adopted child. If both parents are employed by the District, the parents are entitled to one twelve (12) week period of bonding.
- 14.5.3 A unit member may use any accrued sick leave for continuation of pay during the twelve (12) week period of leave. During the twelve (12) week period of leave, the District will continue the District's contribution to medical, dental and vision insurances. If sick leave is exhausted and the employee has remaining differential leave under Section 14.1.3, the District will provide paid differential leave as required by Education Code section 44977.5.

### **14.6** Leave Without Pay for Child Bearing Preparation

- 14.6.1 Leave without pay or other benefits may be granted to a unit member for preparation for child bearing.
- 14.6.2 The unit member shall request such leave as soon as practicable, but under no circumstances less than thirty (30) days prior to the date on which the leave is to

begin. The request shall be in writing and shall include a statement as to the dates the unit member wishes to begin and end the leave without pay.

- 14.6.3 The duration of the leave shall consist of no more than the remainder of the current school year.
- 14.6.4 The unit member is not entitled to the use of any accrued sick leave or other paid leave while on child bearing preparation leave.
- 14.6.5 There shall not be diminution of employment status for child bearing except that no unit member shall be entitled to compensation, increment, nor shall the time taken on child rearing count toward credit for probationary unit members in earning tenure status.
- 14.6.6 In the event of unforeseen circumstances, the unit member may request reassignment to a unit position earlier than the return date of the approved leave. Placement will be contingent upon availability of a vacant positon for which the unit member is qualified, the District will make a concerted effort to assign the unit member as soon as possible and without disruption to the educational process.

## 14.7 Leave Without Pay for Child Rearing

- 14.7.1 Leave without pay or other benefits shall be granted to any unit member who applies for such leave prior to going on pregnancy disability leave, prior to adoption of a child, or in the event of the unit member's spouse's death during childbirth. In all other cases, leave without pay or other benefits may be granted to a unit member for child rearing. The unit member will first have the option of utilizing bonding leave.
- 14.7.2 The unit member shall request such leave as soon as practicable, but under no circumstances less than thirty (30) days prior to the date on which the leave is to begin. The request shall be in writing and shall include a statement as to the dates the unit member wishes to begin and end the leave.
- 14.7.3 The duration of the leave of any unit member with a child (natural or adopted) shall consist of no more than the remainder of the current school year and may be extended for the following school year. An extension of the leave may be granted, not to

exceed an additional twelve (12) months. Child rearing leaves may be granted in accordance with this Section for a unit member with an older child.

- 14.7.4 The unit member is not entitled to the use of any accrued sick leave or other paid leave while on leave for child rearing, except when utilizing bonding leave.
- 14.7.5 There shall not be diminution of employment status for child rearing except that no unit member shall be entitled to compensation or increment, with the exception of time taken for bonding leave, nor shall the time taken on child rearing count toward credit for probationary unit members in earning tenure status.
- 14.7.6 If a unit member is on leave for child rearing, and in the event of the death of the child, the unit member may request an immediate assignment to a unit position. If there is a vacancy for which the unit member is qualified, the District will assign the unit member to a position as soon as possible.

## 14.8 Industrial Accident Leave

- 14.8.1 Unit members will be entitled to industrial accident or illness leave according to the provision in Education Code Section 44984 for personal injury which has qualified for workers' compensation under the provision of the Division of Workers' Compensation and Campbell Union School District's self-insured plan.
- 14.8.2 Industrial accident leave shall not exceed sixty (60) days during which the schools of the District are required to be in session or when the unit member would otherwise have been performing work for the District in any one (1) fiscal year for the same industrial accident.
- 14.8.3 The District has the right to have a unit member examined by a physician designated by the District to assist in determining the length of time during which the unit member will be temporarily unable to perform assigned duties and the degree to which a disability is attributable to the injury involved. All Division of Workers' Compensation Laws of California apply to all employees claiming injury in the course and scope of their employment.

14.8.4 The unit member shall be deemed to have recovered from an industrial accident or illness, and thereby able to return to work, at such time as they and their physician agree that there has been a recovery. The District may require a medical doctor's verification and clearance to return to work.

## 14.9 Judicial Leave

- 14.9.1 Unit members shall be provided leave for regularly called jury duty and to appear as a witness in court, other than as a litigant, for reasons not brought about through the unit member's connivance or misconduct. The unit member shall notify the District immediately upon receipt of notice for jury duty or as a witness.
- 14.9.2 The unit member, while serving jury duty, will receive pay in the amount of the difference between the unit member's regular earnings and any amount received for jury service, excluding expenses.

## 14.10 Legislative Leave

14.10.1 If a permanent unit member is elected to the Legislature, he/she shall be granted a leave of absence from his/her duties as an employee of the District. During the term of such leave of absence, the unit member may be employed by the District to perform less than full-time service requiring certification qualifications, for compensation and upon terms and conditions as may be mutually agreed. The unit member's absence shall not affect in any way their classification. Within six (6) months after the unit member's term of office expires, he/she shall be entitled to return to the position he/she held at the time of his/her election, at the salary to which he/she would have been entitled had he/she not absented himself/herself from the service of the District under this Section. A person employed to take the place of any such unit member shall not have any right to the position following the unit member's return to the position.

## 14.11 Military Leave

14.11.1 Unit members who are members of any reserve corps of the armed forces of the United States or of the National Guard, or who are inducted, or are otherwise ordered

to active military duty shall be granted such leave and military leave pay as is provided in the Military and Veterans Code.

## 14.12 Family and Medical Leave Act

- 14.12.1 Unit members are eligible for leave under the Federal Family and Medical Leave Act (FMLA) and the California Family Rights Act (CFRA). The parties have agreed to the attached memorandum covering the various rights and obligations, including those areas where discretion may be exercised by the District and/or by employees.
- 14.12.2 The provisions of this Agreement and District policies will be applied in conformance with the FMLA and the CFRA.

#### 14.13 Other Leaves Without Pay

- 14.13.1 Upon the Superintendent's recommendation and the Governing Board's approval, leave without compensation, increment, or tenure credit may be granted for a period of one (1) school year. If requested by the unit member, an additional year may be granted upon Governing Board approval.
- 14.13.2 "Other Leaves" shall include, but not be limited to, care for a member of the immediate family who is ill, service in an elected public office, professional study or research, government-sponsored foreign service, temporary relocation of family, seeking alternative career or other family or personal needs.
- 14.13.3 The applications for and granting of such leaves of absence shall be in writing. In addition, a unit member on such leave shall notify the District Human Resources Office by February 1 of the school year of the leave of absence as to their intent to return to District employment. If a unit member fails to provide the District with their intent form by February 1, such failure will constitute a resignation of the position effective June 30 of that year.

#### 14.14 Catastrophic Leave Bank

The Association and the District agree to establish the Campbell Elementary Teachers Association Catastrophic Leave Bank ("the Bank") to assist unit members who suffer from catastrophic illness or injury and have exhausted all available leaves. Unit members who
receive available leave from the Bank shall be referred to as "receiving employees." Unit members shall be allowed to donate and accept eligible leave credits from the Bank according to the following provisions:

14.14.1 Definition

"Catastrophic illness or injury" means an illness or injury that is expected to incapacitate the receiving employee for an extended period of time and which occurs after the receiving employee has exhausted all of his or her paid leaves, including differential (Ed Code 44977 or 45196).

- 14.14.2 Eligibility
  - 14.14.2.1 Unit members who have exhausted all applicable paid leaves provided for in the collective agreement, including differential pay, may apply for Catastrophic Leave under this Article.
  - 14.14.2.2 Any rights to differential pay that receiving employees have under statute or this Agreement shall accrue after the employee has used all accrued paid leave and before receiving catastrophic leave.
  - 14.14.2.3 To qualify for Catastrophic Leave, the unit member must have suffered an illness or injury that is expected to incapacitate him or her for an extended period of time.
  - 14.14.2.4 A unit member's use of Catastrophic Leave shall not exceed a cumulative maximum of sixty (60) days per unit member or the maximum number of days donated pursuant to this section, whichever is less. Catastrophic Leave shall be taken in one (1) day increments.
  - 14.14.2.5 Participation
    - 14.14.2.5.1 Participation in the Catastrophic Leave Bank is voluntary. Unit members who wish to participate will donate at least one (1) sick day annually upon enrollment and one (1) sick day thereafter as needed. New hires will have sixty (60) calendar days from the date of hire to

elect to participate. Only those who donate may receive contributions from the Bank. If a unit member decides to discontinue participation in the Bank, the unit member may not rejoin at a later date.

- 14.14.2.5.2 Open Enrollment shall be from September 1 through October 1, annually.
- 14.14.2.5.3 During Open Enrollment, unit members who have not previously participated in the Bank shall have the opportunity to do so.
- 14.14.2.5.4 Unit members who wish to participate will donate one(1) sick day upon enrollment and one 1) sick daythereafter as needed and determined by the administrating committee.
- 14.14.2.5.5 Should a unit member request to donate to the Bank, but doing so would result in the member having nine (9) or fewer days of accumulated sick leave remaining, the committee shall consider the request on a person-byperson basis.
- 14.14.2.5.6 If a current participant in the Bank declines to donate an additional day requested by the committee, he or she is no longer a participant and my not rejoin the Bank at a later date, except in the circumstances identified in 14.14.2.5.5.

## 14.14.3 Catastrophic Leave Bank Administrating Committee

A Catastrophic Leave Bank Administrating Committee shall be established that includes three (3) unit members appointed by the Association, and one (1) District Office Representative appointed by the Assistant Superintendent of Human Resources.

- 14.14.3.1 The Committee shall be responsible for administering the Catastrophic Leave Bank in accordance with this Agreement and applicable state law. The Committee's duties are:
  - (1) Receiving leave requests;
  - (2) Verifying the validity of requests;
  - (3) Approving or denying requests;
  - (4) Communicating the Committee's decisions to affected unit members, the Human Resources department, and Payroll and
  - (5) Soliciting donations of sick leave from eligible unit members as needed.
- 14.14.3.2 The Committee shall designate one of its members as Chairperson.
- 14.14.3.3 The District shall cooperate with the Committee in establishing appropriate record-keeping procedures, including the total number of accumulated days in the Bank and the names of participating members.
- 14.14.3.4 The Committee shall keep all records confidential and shall not disclose the nature of any illness except as is necessary to process the request for leave and appeals of denials.
- 14.14.3.5 Catastrophic Leave requests shall only be approved by a majority vote of the Committee.
- 14.14.3.6 The Committee will provide Human Resources and Payroll 1) the current bank balance, 2) a list of the prior year's contributors, 3) a list of new enrollees in the Bank and all other contributors for the current year, and, 4) a copy of the letter sent to unit members regarding solicitation, non-solicitation, or status of leave bank donations for the current year. This data shall be provided not later than November 30 of the school year.

#### 14.14.4 Application Procedure

14.14.4.1 A unit member who wishes to use Catastrophic Leave shall submit a request on the appropriate form to the Catastrophic Leave Bank

Committee, stating the facts that support his or her need for Catastrophic Leave. Each request for a withdrawal from the Bank or extension of a withdrawal must be accompanied by medical verification of the need for the leave.

The Committee shall review the unit member's application and make its decision within a reasonable period of time.

- 14.14.4.2 When the Committee determines that the unit member is eligible for Catastrophic Leave, it shall designate the number of days of eligibility. No days may be granted retroactively. The Committee may approve eligibility in renewable increments not to exceed thirty (30) days. If donated days of sick leave are available from the Leave Bank, they may be used by the unit member. If sufficient days are not available, the Committee may solicit donations of days from eligible unit members in accordance with this Agreement on the appropriate form approved by the Association and the District.
- 14.14.4.3 If the Committee denies a request for withdrawal from the Bank, or an extension of withdrawal, because of insufficient days to fund the request or other reasons, the Committee shall notify the unit member, in writing, of the reason for the denial.
- 14.14.4.4 If the Committee reasonably believes that the unit member may be eligible for disability allowance or disability retirement under STRS or Social Security, the Committee may request that the unit member apply for those benefits. If the unit member refuses to submit a complete application, including medical information provided by the unit member's physician, within twenty (20) calendar days, he/she shall no longer be eligible to withdraw days from the Catastrophic Leave Bank for the pending application.

- 14.14.4.5 The unit member shall comply with any requests for additional information from STRS or Social Security within fifteen (15) calendar days, or his/her eligibility to participate in the Catastrophic Leave Bank shall cease. If denied benefits by STRS or Social Security, the unit member must appeal, or his/her eligibility to participate in the Catastrophic Leave Bank shall cease.
- 14.14.5 Donations To Catastrophic Leave Bank
  - 14.14.5.1 Participating unit members shall not be requested to donate more than two(2) days of accrued, full-time sick leave to the Catastrophic Leave Bankin any given school year by the Catastrophic Leave Bank Committee.Donations are irrevocable, and shall not be returned to unit members upontheir cancellation of membership in the Bank.
  - 14.14.5.2 Members of the Catastrophic Leave Bank Committee may solicit an additional day or days, depending upon need, when the bank of days drops to or below fifty (50) days. The solicitation shall be made by the Catastrophic Leave Bank Committee on Campbell Elementary Teachers Association letterhead and shall not be mailed or prepared at District expense. A unit member's membership in the Bank shall be canceled if he or she fails to make a donation when donations are solicited. Eligibility, if available, may only be reestablished in accordance with Section 14.14.2.5.
- 14.14.6 Miscellaneous Provisions
  - 14.14.6.1 Unused days remaining in the Catastrophic Leave Bank shall carry over from year to year.
  - 14.14.6.2 Unit members using days granted to them from the Catastrophic Leave Bank shall not accrue any other leave provided by this Agreement or by law.

- 14.14.6.3 Leave from the Bank may not be used for illness or disability that qualifies a unit member for workers' compensation benefits.
- 14.14.6.4 The Catastrophic Leave Bank is subject to appeal to the Campbell Elementary Teachers Association Executive Board only, and is not subject to review or appeal under any other procedure. Specifically, the Catastrophic Leave Bank Committee's exercise of the rights and discretion described in this Article shall not be subject to the grievance/arbitration procedure. Except for allegations that the District failed to cooperate as required by Section 14.14.3.3 or that the District failed to agree to a form as required by Section 14.14.4.1, no grievance may be filed against the District alleging a violation of this Article.
- 14.14.6.5 If the Catastrophic Leave Bank has insufficient days to fund a withdrawal request, the Committee is under no obligation to provide days and the District is under no obligation to pay the unit member any funds whatsoever.
- 14.14.6.6 If the Catastrophic Leave Bank is terminated for any reason, the days remaining in the Catastrophic Leave Bank shall be returned to the thencurrent members of the Bank proportionately. In no event shall any unit member receive more sick leave days than she/he donated to the Catastrophic Leave Bank.

## 14.15 Miscellaneous

14.15.1 Unless otherwise provided in this Article, a unit member on an unpaid leave of absence may purchase coverage in the District health and welfare programs (i.e., dental, vision, prescription, etc.) at the cost to the District and dependent on condition of the carrier.

## **ARTICLE 15 - ACADEMIC FREEDOM**

## 15.1 Academic Freedom

- 15.1.1 Unit members shall be accorded the right to academic freedom, within the parameters of the State curriculum framework.
- 15.1.2 All instruction shall be fair, accurate, objective, and appropriate to the age and maturity of the student (s), and sensitive to the community needs and the needs and values of our diverse cultures and heritages.

## 15.2 Board Policy

Board Policy Number 6141 shall be maintained during the duration of this Agreement. Before the Board undertakes any modification of this policy, or creates any new policy which impacts upon teacher academic freedom, the District will provide the Association with notice and opportunity to negotiate any such impact.

#### **ARTICLE 16 - TEACHER RIGHTS TO DUE PROCESS**

#### **16.1 General Provisions**

This Article is to establish just cause, due process and progressive discipline for disciplinary action affecting bargaining unit members. The District retains the right to discipline unit members for any reason provided its action is for just cause.

#### **16.2 Progressive Discipline**

The District shall utilize a normal progression: verbal warning; written warning; written reprimand; suspension without pay, in disciplinary actions.

#### 16.2.1 Verbal Warning:

Once the progressive discipline process begins, a verbal warning will be confirmed in writing, which may be in an email. The written confirmation shall not be placed in the member's official personnel file.

#### 16.2.2 Written Reprimand

Written warnings shall not be used unless the unit member has been verbally warned about similar actions within the last four (4) preceding years. The unit member shall sign the reprimand to acknowledge receipt and a copy may be placed in the unit member's personnel file. The unit member has the right to write a response which shall be attached to the reprimand and retained in the file.

#### 16.2.3 Suspension Without Pay for Repeated Offenses

Suspension may be without pay but shall not reduce or deprive the unit member of seniority or any other benefits. No unit member shall receive more than one (1) penalty for any single action or infraction. The suspension shall not exceed fifteen (15) work days. Suspension shall not be used unless the unit member has received a written reprimand about similar actions. No unit member shall be suspended more than fifteen (15) working days during a school year.

#### 16.3 Discipline Without Progression

Nothing in this provision shall prohibit the District from disciplining a unit member for just cause, up to and including suspension without pay for fifteen (15) work days, in instances where remediation is inappropriate.

## **16.4 Procedures** — **Public Charges**

No negative and/or unsatisfactory evaluations shall be based upon information or material of a derogatory or critical nature which has been received by the evaluator from students, parents, and/or citizens unless the procedures described in Section 7.9 have been followed.

## 16.5 Administrative Leave With Pay

The District at its discretion may place any unit member on administrative leave with pay for the purpose of investigating charges or complaints against such unit member. Full benefits and seniority status shall remain in force pending the investigation. Such leave will not be considered disciplinary in nature.

#### **16.6** Notice of Discipline

Notice of suspension shall be made in writing and served in person or by certified mail upon the unit member. A copy shall be provided to the Association President. The notice of suspension shall contain a statement in ordinary language of the specific acts or omissions upon which the action is based, any rule or regulation alleged to have been violated, proposed penalty, and the proposed duration of suspension. The unit member shall be given a copy of any written charges and materials on which the action is based and a statement of the unit member's right to respond.

## **16.7** Appeal of Discipline

- 16.7.1 If the unit member chooses to appeal the imposition of the disciplinary action, the appeal must be taken within five (5) work days from the time of notice of or imposition of the disciplinary action, whichever comes first. The appeal must be made in writing and hand delivered to the office of the Director of Human Resources.
- 16.7.2 Upon appeal the unit member has right to a hearing on the disciplinary action taken by the District. The hearing shall be conducted by the Superintendent or designee

within ten (10) work days of the receipt of the notice of appeal. A record of the hearing shall be made by the District through electronic recordation, or, if mutually agreed by both parties, a court reporter's transcription shall be utilized. The Superintendent or designee shall provide a written decision within five (5) work days of the completion of the hearing.

## 16.8 Arbitration

The Association may appeal the decision of the Superintendent directly to arbitration pursuant to Article 5 of this Agreement.

- 16.8.1 At the arbitration, documentation supporting discipline, including written reprimands, may be subject to the arbitrator's review.
- 16.8.2 Evidence will not be admitted at any level of this process which supports events which occurred more than four years prior to the incident(s) leading to discipline; except where the evidence is submitted to prove a pattern of behavior which is similar to the incident(s) cited as the basis for discipline.

## **16.9** Association Presence

The Association, as the exclusive representative, has the right to be present at all conferences and hearings, regardless of any request by the unit member who is the subject of the disciplinary action.

## **16.10 Effect of Pending Appeal**

If an appeal is filed by the unit member or the Association related to the discipline of the unit member, then all suspension actions proposed by the District shall be stayed pending a final decision on the appeal.

#### **16.11** Confidentiality

Unless agreed otherwise by the parties, the information and proceedings relative to such disciplinary actions shall remain confidential with the individuals, unit members, and parties who are involved, and shall not be released without the consent of the Association and the District, except as required by law.

## 16.12 Limitations

Nothing in the above provisions will be construed to limit the rights of the District as set forth in Article 3, except as limited by the specific and precise language of this Section.

## 16.13 Post Discipline

If, after having been disciplined, a unit member serves the District for twelve (12) months without the need for further disciplinary action, the Association and the unit member shall be given a follow-up notice to that effect which shall also be attached to any original notice that may have been placed in the unit member's personnel file.

## 16.14 Application

This Article is not intended to apply to suspensions pursuant to Education Code Sections 44939, 44940, or 44942.

#### **ARTICLE 17 - RETIREMENT PROVISIONS**

A unit member with the equivalent to ten (10) years of full-time service in the District who has reached the required age may request in writing to participate in one of the following retirement plans. A request to participate in an early retirement plan must be submitted to the Human Resources Office no later than March 1 of the school year preceding the desired year of participation. In addition, these retirement incentives shall remain effective only as long as the District experiences a net savings when comparing the cost of a replacement teacher to the cost of the retiring teachers.

## **17.1 Pre-Retirement/Part-Time Employment**

## 17.1.1 Eligibility Requirements

The District shall allow unit members to be employed on a part-time basis for early retirement purposes. In order to participate in this program, the certificated unit member must meet the following criteria:

- 17.1.1.1 The option of part-time employment must be exercised at the unit member's request and, once granted, can be revoked only with the mutual consent of the District and the unit member. The provisions of the program can be increased or decreased with the code limitations.
- 17.1.1.2 The unit member must submit a request to participate in this program to the Human Resources Office no later than March 1 of the school year preceding the desired year of participation.
- 17.1.1.3 The unit member must have reached the age of fifty-five (55) prior to the school year or term in which the reduction in work load starts.
- 17.1.1.4 The unit member must have been employed full-time in a position requiring certification for at least ten (10) years, of which the immediately preceding five (5) years were full-time employment.
- 17.1.1.5 Unit members may not participate after age sixty-five (65). Individuals in the program who reach age sixty-five (65) during the school year may continue through the school year.

- 17.1.1.6 The District and the unit member shall agree to make the appropriate contributions to the State Teachers' Retirement System (STRS) equal to the amount required as if serving as a full-time unit member.
- 17.1.1.7 The final determination as to which unit members will participate in this program and the form of part-time employment rests within the sole discretion of the Governing Board.
- 17.1.2 The minimum part-time employment shall be the equivalent of one-half (1/2) of the number of days of service required by the unit member's contract of employment during his/her final year of service in a full-time position. Except for the reduction of salary corresponding to the reduced workload, the District will provide the part-time unit member with the same fringe benefits provided a regular full-time unit member.
- 17.1.3 If the Governing Board agrees, the reduced service may be on a daily schedule or full-time for at least one-half (1/2) year; however, the unit member and District contributions must be paid monthly to STRS.
- 17.1.4 A member of the State Teachers' Retirement System employed on a part-time basis under the provisions of this policy shall receive the credit he/she would receive if he/she was employed on a full-time basis and have his/her retirement allowance, as well as any other benefits that he/she is entitled to under the Education Code, based upon the salary that he/she and District both elect to contribute to the Teachers' Retirement Fund the amount that would have been contributed if the member was employed on a full-time basis. This provision shall be applicable only to members of the STRS who have met the criteria provided in the Education Code and are not older than sixty-five (65) years, and is limited to a period of five (5) years of such part-time status.
- 17.1.5 Unit members who terminate before the end of the school term or the school year will receive retirement credit based on the salary actually paid in the proportion that it relates to the annual salary that would have been paid had the employment

continued. Retirement contributions for services not actually performed will be returned to the unit member and the District. If a unit member works less than one hundred percent (100%) of his/her assignment, the amount of sick leave earned varies directly to the percent of full-time employment. That is, fifty percent (50%) employment would yield ten (10) days of half-time sick leave or five (5) days of full-time sick leave.

17.1.6 A unit member on less than full-time employment, who must use sick leave, will reduce earned sick leave on the same basis as his/her employment.

#### **17.2** Early Retirement/Consultant Contract

Early retirees who meet the retirement criteria set forth in Education Code Section 35046 may enter into a consultancy contract, an agreement with the District to provide services to the District. The District retains the right to determine the person(s) selected, the length of service, and the nature of the service. Early retirees who are selected will be paid at a rate of one hundred fifty dollars (\$150) per day or a fractional part of that amount not to exceed seven thousand five hundred dollars (\$7,500) per year for services rendered. Retirees may also work as substitutes for the District and shall be paid twenty-five dollars (\$25) per day in addition to the prevailing substitute rate.

#### **17.3** Retiree Benefits

17.3.1 <u>Conditions</u>

The following conditions must be met in order to receive benefits pursuant to Section 17.3.2 or 17.3.3:

- 17.3.1.1 Retirement commencing after age fifty-five (55) shall be purely voluntary on the part of the unit member.
- 17.3.1.2 A unit member must have a minimum of ten (10) years of service in the Campbell Union School District in a position requiring certification.

- 17.3.1.3 The unit member must permanently resign from the District in order to be eligible for this program. A unit member who returns to the District pursuant to Section 17.5 remains eligible for this program.
- 17.3.2 The District will contribute up to three thousand two hundred and fifty dollars (\$3,250) per year toward the cost of the employee and spouse coverage in a medical and dental health plan benefit program offered by the District. The District's contribution of up to three thousand two hundred and fifty dollars (\$3,250) shall continue for seven (7) years, or until employee is eligible for health benefits under Federal of State programs (i.e., Medicare, Medi-Cal), whichever comes first. If there are changes in health and welfare fringe benefit carriers, participants in the early retirement plan will also be changed to the new carrier provided the new carrier will accept the risk. If the employee becomes eligible for health benefits under Federal or State programs and has not received a total of seven (7) years of District contribution toward the cost of retirees medical and dental plans, the District shall contribute up to one thousand seven hundred dollars (\$1,700) per year toward the cost of Medicare or Medi-Cal supplemental coverage and/or a dental plan until the employee has received a total of seven (7) years of contributions. If there are changes in health and welfare fringe benefit carriers, participants in the early retirement plan will also be changed to the new carrier provided the new carrier will accept the risk.

#### **17.4** Retirement for District Determined Disability

- 17.4.1 The District will establish a special annuity program for unit members who the District determines to suffer from a bona fide partial disability which has substantially reduced prior effectiveness in the classroom.
- 17.4.2 Such disability shall be demonstrated through District selected medical examinations and other proof as may be required by the Governing Board. The reduced effectiveness in the classroom shall be established by the District.
- 17.4.3 Eligibility for this program shall require:

- (1) initiation by the unit member;
- (2) at least twenty (20) years of service to the District;
- (3) at least fifty-five (55) years in age; and
- (4) a willingness to apply and be accepted for retirement under STRS.
- 17.4.4 In granting an annuity, the District will not expend more than ten thousand dollars (\$10,000) per unit member. The annuity shall be obtained either through
- 17.4.5 private sources or through the STRS. The granting of this benefit and the determination of the amount of benefits shall be at the Governing Board's total discretion.

#### 17.5 Post Retirement Employment Program

- 17.5.1 The District may employ in a full-time teaching position a teacher who retired from the District under the State Teachers Retirement System ("STRS") and who meets either of the following:
  - 17.5.1.1 The teacher retired with an effective date on or before January 1, 2000, and will provide direct classroom instruction to students in transitional

kindergarten

through eighth grade, and/or will provide services to beginning teachers specified in Education Code Section 24216.5(a)(2); or

- 17.5.1.2 The teacher retired with an effective date on or before July 1, 2000, and will provide direct remedial instruction to students in grades 2 through 8 as defined in Education Code Sections 37252 and 37252.5. Subsection 17.5.1.2 will become effective on January 1, 2001.
- 17.5.2 Retired teachers employed pursuant to this program shall be placed in distinct classes of temporary teachers within the bargaining unit. A teacher shall be classified as a "Retired Temporary Teacher" if hired pursuant to section 17.5.1.1 and as a "Retired Temporary Remedial Teacher" if hired pursuant to Section 17.5.1.2. The service of a Retired Temporary or a Retired Temporary Remedial Teacher shall not be included

in computing the service required as a prerequisite to attainment of or eligibility for classification as a permanent employee of a school district.

- 17.5.3 Retired Temporary Teachers and Retired Temporary Remedial Teachers shall be compensated according to the salary schedule set forth in Appendix A.
- 17.5.4 Retired Temporary Teachers and Retired Temporary Remedial Teachers shall not receive health and welfare benefits pursuant to Article 10, Section 10.5 of this Agreement, but instead shall continue to receive the retiree benefit contributions for these teachers shall not be extended beyond those specified in Sections 17.3.2 and 17.3.3.
- 17.5.5 Retired Temporary Teachers and Retired Temporary Remedial Teachers shall not be subject to the evaluation requirements of Article 17.5.

## **17.6** Concurrent Participation

Members may not concurrently participate in both the programs specified in Sections 17.1 or 17.2 and the post-retirement program specified in Section 17.5

## ARTICLE 18 - WAIVER PROCESS FOR CHANGING BARRIERS TO SITE-BASED SHARED RESPONSIBILITY AND DECISION MAKING (SBSRDM)

#### 18.1 SBSRDM

In appropriate circumstances, SBSRDM will be used in the District. Barriers or limitations to this process will be removed by using a case-by-case waiver of those contributing forces, which include Board Policy, collective agreements, the Education Code, and Central Services. The parties agree to adopt the following procedure for waiving provisions of this collective bargaining agreement.

## 18.2 Procedure

The request for a waiver should be forwarded to the District Superintendent and CETA president who shall seek approval of the waiver according to the provisions of 18.3. If either the Association or the District request verification, the site shall submit verification that by a 2/3 vote the staff approved the waiver. A waiver request shall include at least the following:

- 18.2.1 The specific provision of the collective bargaining agreement to be waived.
- 18.2.2 The replacement language which will be in effect in place of the waived language.
- 18.2.3 All unit members (and positions) who would be affected by the change, and a statement that by a 2/3 vote these members agreed to the change.

## **18.3** Waiver Approval

If the waiver involves an adjustment to a collective agreement, the waiver does not become effective until the adjustment is approved in writing by CETA and the Board. If the waiver involves a change in Board Policy, the Board must approve. If the waiver involves an alteration of Central Services, the Superintendent must approve. If the waiver involves the Education Code, the established procedure for Education Code waivers will be utilized.

## **18.4 Record of Waivers**

All waivers that are approved by the appropriate parties will be reduced to writing and numbered for reference.

#### **ARTICLE 19 - SPECIAL EDUCATION**

#### **19.1** Release days for impactful caseloads

Release days may be available to Special Education teachers who request days for the purpose of managing significantly impactful caseload issues. The Special Education Department shall determine whether the days are needed. The teacher shall select dates from the dates provided by the Department and confirm a date with the Special Education office. Approval is required.

#### **19.2** Non-work days

If the release days are needed when there is no available coverage, scheduled meetings, or other site or District training activities, Special Education teachers are entitled to up to 2 non-work days per year for their necessary duties. To use a non-work day, the Special Education teacher shall obtain prior written approval from their Supervisor. The unit member(s) shall be compensated at one and a half the rate in section 10.6.2 for 6.5 (six and a half) hours per non-work day.

## **19.3** Prep time

Special Education teachers shall not be pulled from their prep period to attend to their own student(s)' needs, unless the teacher agrees or the administrator directs the teacher to provide assistance. In the rare instance when a teacher is pulled by an administrator from his or her prep, the prep time shall be made up or the teacher will be paid for that period at one and a half the rate in section 10.6.2.

#### **19.4** SDC support for more than 3 grade levels

Within the first 30 student contact days of the school year, the members with three (3) or more grade levels, in their Mild to Moderate SDC classrooms shall participate in a meeting with the Special Education Director or designee to determine if support and assistance may be needed to help make the delivery of multiple grade levels instruction more effective. A report of the meetings, participants, issues discussed and supports to be implemented shall be generated and provided to CUSD and CETA. Additional meetings, with the SDC teacher and the Special Education Director or designee shall occur once per trimester, to determine if the initial needs have changed or supports are successful.

#### **ARTICLE 20 - MISCELLANEOUS PROVISIONS**

#### 20.1 Effects of Agreement

The terms and conditions set forth in this Agreement represent the full and complete understanding between the parties. Except as provided for in Section 19.6.1, the Association and the District expressly waive the right to meet and negotiate and agree that neither the District nor the Association shall be obligated to meet and negotiate with respect to any subject covered in this Agreement, unless modified through the voluntary, mutual consent of the parties in a written amendment. This Agreement terminates and supersedes those partial practices, agreements, procedures, traditions, and rules or regulations inconsistent with any matters covered in this Agreement. The parties agree that during the negotiations which culminated in this Agreement, each party enjoyed the opportunity to make demands and proposals or counter-proposals with respect to any matter even though some matters were proposed and later withdrawn, and that the understandings and agreements arrived at after the exercise of that right and opportunity are executed in this Agreement.

#### 20.2 Individual Contracts

Any individual contract between the District and an individual unit member shall be subject to and consistent with the terms and conditions of this Agreement. If an individual contract contains any language inconsistent with this Agreement, this Agreement during its duration shall be controlling.

#### 20.3 Savings

Should any Article, Section, or clause of this Agreement be declared illegal by a court of competent jurisdiction, that Article, Section, or clause, as the case may be, shall be automatically deleted from this Agreement to the extent that it violated the law. The remaining unaffected Articles, Sections, and clauses shall remain in full force and effect for the duration of the Agreement.

#### 20.4 Support of Agreement

The District and the Association agree that it is to their mutual benefit to encourage the resolution of differences through the meet and negotiation process. Therefore, it is agreed

that the Association will support this Agreement for its term and will not appear before the Governing Board to seek change or improvement in any matter contained in this Agreement, except by mutual agreement of the District and the Association, or as provided for in this Agreement.

## 20.5 Concerted Activities

- 20.5.1 The Association, its officers, agents or unit members will not strike, engage in a work stoppage, slowdown, concerted refusal to perform job functions and responsibilities, or picketing in furtherance thereof, nor will they comply with the request of other labor organization(s) to engage in such activity, nor engage in any unlawful interference with the District's operation.
- 20.5.2 The Association recognizes the duty and obligation of its representatives to comply with the provisions of this Agreement and to make every effort toward inducing all unit members to do so.
- 20.5.3 Provided there is no violation of this Article 19, the District will not lockout the bargaining unit members.
- 20.5.4 This provision shall not restrict either party from seeking redress for an alleged violation of this Article through Public Employment Relations Board procedures or a court of competent jurisdiction.

#### 20.6 Term of Agreement

The term of this agreement shall be 2022-2025. This Agreement shall take effect on the date of signing by the parties except as specifically described in any Article, and shall remain in full force and effect, up to and including June 30, 2025, and thereafter shall continue in effect year-to-year unless one of the parties notifies the other in writing no later than April 15, of its request to modify, amend, or terminate this Agreement.

20.6.1 For the second and third year of this Agreement the parties agree to reopen on Article 10 and one article selected by each party.

The tentative agreement between the Campbell Union School District and the Campbell Elementary Teachers Association closes all bargaining for 2022-2023 and 2023-2024, including all impacts of bargaining for 2024-2025.

## **20.7 Printing the Agreement**

The cost of printing the collective agreements for CETA will be equally shared by the District and the Association.

FOR THE DISTRICT 1

FOR CET ane

## **APPENDIX A - SALARY SCHEDULE**

			Salary Schedule		
	Column A	Column B	Column I	Column II	Column III
STEP (Years of Experience)	No credential (intern permit, waiver, etc.)	Credential and BA + 29 or fewer SEMESTER UNITS	Credential and BA + 30 - 44 SEMESTER UNITS	Credential and BA + 45 - 59 SEMESTER UNITS	Credential and BA + 60 or more SEMESTER UNITS
1	\$73,342	\$74,836	\$74,839	\$75,465	\$76,096
2			\$75,465	\$76,096	\$78,287
3			\$76,096	\$76,860	\$81,444
4			\$76,299	\$80,918	\$85,494
5			\$79,149	\$84,073	\$89,548
6			\$83,196	\$88,124	\$93,597
7			\$87,244	\$92,174	\$97,648
8			\$91,293	\$96,222	\$101,694
9			\$95,347	\$100,276	\$105,753
10			\$99,394	\$104,325	\$109,795
11			\$103,453	\$108,374	\$113,849
12				\$112,422	\$117,899

#### CAMPBELL UNION SCHOOL DISTRICT CETA Annual Salary Schedule

Supermaximum (Column III only)

STEP	Annual Stipend	Total w/base
15	\$2,500	\$120,399
20	\$2,250	\$122,649
24	\$2,500	\$125,149
25	\$2,500	\$127,649

Annual Stipends:

Master's Degree or Education Specialist (EdS) Stipend - \$2,800

Supermaximum (longevity) - as outlined above

Annual Health Coverage Benefit:

District Contribution - \$12,203

Other Information:

185 day work year

Maximum of 10 years credit for outside experience

Board Approved: 09/29/22 Salary Rates Effective: 07/01/22 Benefit Rates Effective 01/01/22 Sick Leave Accrual:

6.5 hours per work month

CETA 22-23 (9%)

## **APPENDIX B – CALENDAR**

#### Campbell Union School District

Employee Work Year Calendar

2022-2023

M T W Th F 4 Independence Day Observed (legal	
	holiday) M T W Th F
1 25 11 mo. Administrators return	1 2 3 4 5
4 5 6 7 8 <b>28</b> 11 mo. Classified Staff return	8 9 10 11 12
11 12 13 14 15 Aug	ust 15 16 17 <u>18</u> 19
18         19         20         21         22         15         Professional Development Day & Back to S	ichooltrainingsforal Staff 22 23 24 25 26
25 26 27 28 29 16, 17 CETA Work Days (non-work day for	10 mo. Classified staff) 29 30 31
<u>18</u> First Day of School	
SEPTEMBER September	mber OCTOBER
M T W Th F 5 Labor Day (legal holiday)	M T W Th F
1 2 Octo	ber ::3: :4: :5: :6: :7:
5 6 7 8 9 Parent-Teacher Conference Days (1s	st report card) (shortened days for students) 10 11 12 13 14
12   13   14   15   16   10   Professional Development Day (non	n-work day for 10 & 11 mo. Classified staff) 17 18 19 20 21
19 20 21 22 23 Nover	mber 24 25 26 27 28
26     27     28     29     30     11     Veterans Day (legal holiday)	31
21, 22 Thanksgiving break (non-work days	
NOVEMBER 24 Thanksgiving Day (legal holiday)	DECEMBER
M T W Th F 23, 25 Thanksgiving break (District holiday	
	5         6         7         8         9           12         13         14         15         16
14         15         16         17         18         23         Winter Break (non-work day)           21         22         23         24         25         19-21, 27, 28         Winter Break (CSEA mandatory vaca	
21         22         23         24         25         1922, 21, 23         winter break (CSDA mandalofy value)           28         29         30         26         Christmas Day (legal holiday) obset	
JANUARY 2 New Years (observed)	FEBRUARY
M T W Th F 16 Martin Luther King Jr.'s Birthday	/ (legal holiday) M T W Th F
2 3 4 5 6 Febr	1 2 3
9 10 11 12 13 6-10 Intervention Conferences	<u>6</u> <u>7</u> <u>8</u> <u>9</u> <u>10</u>
16         17         18         19         20         20         Washington's Birthday (legal holid	lay) observed 13 14 15 16 17
23 24 25 26 27 21-23 Presidents' Week break (non-work i	days - 10 mo. staff) 20 21 22 23 24
30 31 24 Lincoln's Birthday (legal holiday) o	bbserved 27 28
Mar	
MARCH 10 Professional Development Day (non	
M T W Th F Ap	
1 2 3 10-14 Spring Break (CSEA mandatory vaca	
6         7         8         9         10         24         SBAC Window Opens           13         14         15         16         17         Ma	<b>10</b> 11 12 13 14 <b>av</b> 17 18 19 20 21
20         21         22         23         24         26         SBAC Window Closes           27         28         29         30         31         29         Memorial Day (legal Holiday)	<u>24</u> 25 26 27 28
27 28 29 30 31 29 Memorial Day (legal Holiday)	
MAY <u>9</u> Last day of School (last work day fo	
M T W Th F 16 Last work day for 11 mo. Classifier	
1 2 3 4 5 19 Juneteenth	
8 9 10 11 12 27 Last work day for 11 mo. Administ	
	12 13 14 15 16
15 16 17 18 19 Shortened Shortened Days for students.	
15         16         17         18         19         Shortened         Shortened Days for students.           22         23         24         25         26         Day         Afternoons are reserved for employ	

2022-2023

## Campbell Union School District

Employee Work Year Calendar

2023-2024

		JULY				ylul		P	UGUS	т	
м	т	w	Th	F	4	Independence Day (legal holiday)	м	Т	w	Th	F
3	4	5	6	7	24	11 mo. Administrators return		1	2	3	4
10	11	12	13	14	26	11 mo. Classified Staff return	7	8	9	10	11
17	18	19	20	21		August	14	15	16	17	18
24	25	26	20	28	10		21	22	23	24	25
_	25	20	27	20		Professional Development Day (non-work day for 10 mo. Classified staff) Professional Development Day & Back to School trainings for ALL Staff	21	22	30	31	25
31					11		28	29	50	51	
				25722	CETA Work Days(non-work day for 10 mo. Classified staff)	OCTOBER					
	1117	10.00	20.0152	_	<u>16</u>	First Day of School		142.02	10.00	ACCENT	-
м	Т	W	Th	F		September	M	T	W	Th	F
1000				1	4	Labor Day (legal holiday)	2	3 :	4	5	6
4	5	6	7	8		October	9	10	11	12	13
11	12	13	14	15	2-6	Parent-Teacher Conference Day{1st report card)(shortened days for students)	16	17	18	19	20
18	19	20	21	22	20	Professional Development Day (non-work day for 10 & 11 mo. Classified staff)	23	24	25	26	27
25	26	27	28	29		November	30	31			
-					10	Veterans Day observed (legal holiday)			II INCLUSION		
	NC	VEME	BER		20, 21	Thanksgiving break (non-work days - 10 & 11 mo. staff)		DE	CEMB	ER	
м	T	w	Th	F	23	Thanksgiving Day (legal holiday)	М	T	W	Th	F
		1	2	3	22, 24	Thanksgiving break (District holidays)					1
6	7	8	9	10		December	4	5	6	7	8
13	14	15	16	17	22, 28, 29	Winter Break (District holidays)	11	12	13	14	15
20	21	22	23	24	18	Winter Break (non-work day - 10 & 11 mo. staff)	18	19	20	21	22
27	28	29	30		19-21, 26, 27	Winter Break (CSEA mandatory vacation days)	25	26	27	28	29
					1075/10	to be the boot offered on the sec	3				
JÁNUARY					25	Christmas Day (legal holiday) observed					
	J,	NUA	RY		25	Christmas Day (legal holiday) observed January		FE	BRUA	RY	
М	ע ד	NUAI W	RY Th	F	25		м	FE T	BRUA W	RY <i>Th</i>	F
M 1		Contra v Portos		F 5		January	М			10000	F 2
1942530	Т	w	Th	<u>.</u>	1	January New Years	M 5			Th	<u></u>
1	T 2	W 3	Th 4	5	1	January New Years Martin Luther King Jr.'s Birthday (legal holiday)		т	w	<i>Th</i> 1	2
1 8	T 2 9	W 3 10	Th 4 11	5 12	1 15	January New Years Martin Luther King Jr.'s Birthday (legal holiday) February	<u>5</u>	Т <u>б</u>	w Z	Тһ 1 <u>8</u>	2 <u>9</u>
1 8 15	T 2 9 16	W 3 10 17	<i>Th</i> 4 11 18	5 12 19	1 15 <u>5-9</u>	January New Years Martin Luther King Jr.'s Birthday (legal holiday) February Intervention Conferences	<u>5</u> 12	Т <u>6</u> 13	W Z 14	Th 1 <u>8</u> 15	2 <u>9</u> 16
1 8 15 22	Т 2 9 16 23	W 3 10 17 24	<i>Th</i> 4 11 18	5 12 19	1 15 <u>5-9</u> 19	January New Years Martin Luther King Jr.'s Birthday (legal holiday) February Intervention Conferences Washington's Birthday (legal holiday) observed	<u>5</u> 12 19	т <u>6</u> 13 20	W Z 14 21	Th 1 8 15 22	2 <u>9</u> 16
1 8 15 22	T 2 9 16 23 30	W 3 10 17 24	Th 4 11 18 25	5 12 19	1 15 <u>5-9</u> 19 20-22	January New Years Martin Luther King Jr.'s Birthday (legal holiday) February Intervention Conferences Washington's Birthday (legal holiday) observed Presidents' Week break (non-work days - 10 mo. staff)	<u>5</u> 12 19	т <u>6</u> 13 20 27	W Z 14 21	Th 1 8 15 22	2 <u>9</u> 16
1 8 15 22	T 2 9 16 23 30	W 3 10 17 24 31	Th 4 11 18 25	5 12 19	1 15 <u>5-9</u> 19 20-22	January New Years Martin Luther King Jr.'s Birthday (legal holiday) February Intervention Conferences Washington's Birthday (legal holiday) observed Presidents' Week break (non-work days - 10 mo. staff) Lincoln's Birthday (legal holiday) observed	<u>5</u> 12 19	т <u>6</u> 13 20 27	W Z 14 21 28	Th 1 8 15 22	2 <u>9</u> 16
1 8 15 22 29	T 2 9 16 23 30	W 3 10 17 24 31	Th       4       11       18       25	5 12 19 26	1 15 <u>5-9</u> 19 20-22 23	January New Years Martin Luther King Jr.'s Birthday (legal holiday) February Intervention Conferences Washington's Birthday (legal holiday) observed Presidents' Week break (non-work days - 10 mo. staff) Lincoln's Birthday (legal holiday) ob served March	<u>5</u> 12 19 26	т <u>6</u> 13 20 27	W 2 14 21 28 APRIL	Th         1 <b>8</b> 15         22         29	2 <u>9</u> 16 23
1 8 15 22 29	T 2 9 16 23 30	W 3 10 17 24 31	Th       4       11       18       25	5 12 19 26 <i>F</i>	1 15 <u>5-9</u> 19 20-22 23	January         New Years         Martin Luther King Jr.'s Birthday (legal holiday)         February         Intervention Conferences         Washington's Birthday (legal holiday) observed         Presidents' Week break (non-work days - 10 mo. staff)         Lincoln's Birthday (legal holiday) ob served         March         Optional Professional Development Day (non-work day for 10 & 11 mo. staff) Admin work day	5 12 19 26 <i>M</i>	Т <u>6</u> 13 20 27 Т	W <u>Z</u> 14 21 28 APRIL W	Th         1         8         15         22         29         Th	2 9 16 23 <i>F</i>
1 8 15 22 29 <i>M</i> 4	Т 9 16 23 30 Т 5	W 3 10 17 24 31 WARCI W 6	Th         4         11         18         25         Th         Th         7	5 12 19 26 <i>F</i> 1 8	1 15 <u>5-9</u> 19 20-22 23 15	January         New Years         Martin Luther King Jr.'s Birthday (legal holiday)         February         Intervention Conferences         Washington's Birthday (legal holiday) observed         Presidents' Week break (non-work days - 10 mo. staff)         Lincoln's Birthday (legal holiday) ob served         March         Optional Professional Development Day (non-work day for 10 & 11 mo. staff) Admin work day         April         Spring Break (CSEA mandatory vacation days)	5 12 19 26 <i>M</i> 1 8	Т 13 20 27 Т 2 9	W 2 14 21 28 APRIL W 3 10	Th         1         22         29         Th         4         11	2 9 16 23 <i>F</i> 5 12
1 8 15 22 29 <i>M</i> 4 11	T         2         9         16         23         30         T         5         12	W           3           10           17           24           31           WARCI           W           6           13	Th         4         11         18         25         Th         7         14	5 12 19 26 <i>F</i> 1 8 15	1 15 <u>5-9</u> 19 20-22 23 15	January         New Years         Martin Luther King Jr. 's Birthday (legal holiday)         February         Intervention Conferences         Washington's Birthday (legal holiday) observed         Presidents' Week break (non-work days - 10 mo. staff)         Lincoln's Birthday (legal holiday) ob served         March         Optional Professional Development Day (non-work day for 10 & 11 mo. staff) Admin work day         April         Spring Break (CSEA mandatory vacation days)         SBAC Window Opens	5 12 19 26 <i>M</i> 1	Т <u>6</u> 13 20 27 7 2 9 16	W           Z           14           21           28           APRIL           W           3           10           17	Th         1 <u>8</u> 15         22         29         Th         4         11         18	2 9 16 23 <i>F</i> 5 12 19
1 8 15 22 29 <i>M</i> 4 11 18	T           2           9           16           23           30           T           5           12           19	W           3           10           17           24           31           WARCO           Ø           6           13           20	Th           4           11           18           25	5 12 19 26 <i>F</i> 1 8 15 22	1 15 <u>5-9</u> 19 20-22 23 15 8-12 <u>22</u>	January         New Years         Martin Luther King Jr.'s Birthday (legal holiday)         February         Intervention Conferences         Washington's Birthday (legal holiday) observed         Presidents' Week break (non-work days - 10 mo. staff)         Lincoln's Birthday (legal holiday) ob served         March         Optional Professional Development Day (non-work day for 10 & 11 mo. staff) Admin work day         April         Spring Break (CSEA mandatory vacation days)         SBAC. Window Opens         May	5 12 19 26 <i>M</i> 1 8 15 22	Т <u>6</u> 13 20 27 Т 2 9 16 23	W 2 14 21 28 APRIL W 3 10	Th         1         22         29         Th         4         11	2 9 16 23 <i>F</i> 5 12
1 8 15 22 29 <i>M</i> 4 11	T         2         9         16         23         30         T         5         12	W           3           10           17           24           31           WARCI           W           6           13	Th         4         11         18         25         Th         7         14	5 12 19 26 <i>F</i> 1 8 15	1 15 <u>5-9</u> 19 20-22 23 15 <u>8-12</u> <u>22</u> <u>24</u>	January         New Years         Martin Luther King Jr.'s Birthday (legal holiday)         February         Intervention Conferences         Washington's Birthday (legal holiday) observed         Presidents' Week break (non-work days - 10 mo. staff)         Lincoln's Birthday (legal holiday) ob served         March         Optional Professional Development Day (non-work day for 10 & 11 mo. staff) Admin work day         April         Spring Break (CSEA mandatory vacation days)         SBAC. Window Opens         May         SBAC. Window Closes	5 12 19 26 <i>M</i> 1 8	Т <u>6</u> 13 20 27 7 2 9 16	W           Z           14           21           28           APRIL           W           3           10           17	Th         1 <u>8</u> 15         22         29         Th         4         11         18	2 9 16 23 <i>F</i> 5 12 19
1 8 15 22 29 <i>M</i> 4 11 18	T           2           9           16           23           30           T           5           12           19	W 3 10 17 24 31 31 WARC( W 6 13 20 27	Th           4           11           18           25	5 12 19 26 <i>F</i> 1 8 15 22	1 15 <u>5-9</u> 19 20-22 23 15 8-12 <u>22</u>	January         New Years         Martin Luther King Jr.'s Birthday (legal holiday)         February         Intervention Conferences         Washington's Birthday (legal holiday) observed         Presidents' Week break (non-work days - 10 mo. staff)         Lincoln's Birthday (legal holiday) ob served         March         Optional Professional Development Day (non-work day for 10 & 11 mo. staff) Admin work day         April         Spring Break (CSEA mandatory vacation days)         SBAC Window Closes         Maponical Day (legal holiday)	5 12 19 26 <i>M</i> 1 8 15 22	Т <u>6</u> 13 20 27 Т 2 9 16 23	W           Z           14           21           28           APRIL           W           3           10           17           24	Th         1 <u>8</u> 15         22         29         Th         4         11         18	2 9 16 23 <i>F</i> 5 12 19
1 8 15 22 29 <i>M</i> 4 11 18 25	T         2         9         16         23         30         T         5         12         19         26	W           3           10           17           24           31           MARCO           6           13           20           27           MAY	Th         4         11         18         25         7         14         21         28	5 12 19 26 7 1 8 15 22 29	1 15 <u>5-9</u> 19 20-22 23 15 8-12 22 22 22 22 23	January         New Years         Martin Luther King Jr.'s Birthday (legal holiday)         February         Intervention Conferences         Washington's Birthday (legal holiday) observed         Presidents' Week break (non-work days - 10 mo. staff)         Lincoln's Birthday (legal holiday) ob served         March         Optional Professional Development Day (non-work day for 10 & 11 ma. staff) Admin work day         April         Spring Break (CSEA mandatory vacation days)         SBAC Window Opens         May         SBAC Window Closes         Memorial Day (legal holiday)         June	5 12 19 26 <i>M</i> 1 8 15 22 29	Т <u>6</u> 13 20 27 Т 2 9 16 23 30	W           Z           14           21           28           APRIL           W           3           10           17           24	Th           1 <u>8</u> 15           22           29           Th           4           11           18           25	2 9 16 23 5 12 19 26
1 8 15 22 29 <i>M</i> 4 11 18	T           2           9           16           23           30           T           5           12           19	W 3 10 17 24 31 WARC W 6 13 20 27 WAY W	Th           4           11           18           25           Th           7           14           21           28           Th	5 12 19 26 <i>F</i> 1 8 15 22 29 <i>F</i>	1 15 <u>5-9</u> 19 20-22 23 15 8-12 22 22 24 27 <u>6</u>	January         New Years         Martin Luther King Jr.'s Birthday (legal holiday)         February         Intervention Conferences         Washington's Birthday (legal holiday) observed         Presidents' Week break (non-work days - 10 mo. staff)         Lincoh's Birthday (legal holiday) ob served         March         Optional Profe ssional Development Day (non-work day for 10 & 11 mo. staff) Admin work day         April         Spring Break (CSEA mandatory vacation days)         SBAC Window Opens         May         SBAC Window Closes         Memorial Day (legal holiday)         June         Last day of School (last work day for 10 mo. staff)	5 12 19 26 <i>M</i> 1 1 8 15 22 29 <i>M</i>	Т <u>6</u> 13 20 27 7 2 9 16 23 30 Т	W Z 14 21 28 APRIL W 3 10 17 24 JUNE W	Th           1           g           15           22           29           Th           4           11           18           25           Th	2 9 16 23 <i>F</i> 5 12 19 26 <i>F</i>
1 8 15 22 29 <i>M</i> 4 11 18 25 <i>M</i>	T 2 9 16 23 30 T 5 12 19 26 T	W           3           10           17           24           31           WARCO           W           6           13           20           27           WAY           1	Th         4         11         18         25         Th         7         14         21         28         Th         2	5 12 19 26 <i>F</i> 1 8 15 22 29 <i>F</i> 3	1 15 5-9 19 20-22 23 15 8-12 22 24 27 24 27 6 13	January         New Years         Martin Luther King Jr.'s Birthday (legal holiday)         February         Intervention Conferences         Washington's Birthday (legal holiday) observed         Presidents' Week break (non-work days - 10 mo. staff)         Lincoh's Birthday (legal holiday) observed         March         Optional Professional Development Day (non-work day for 10 & 11 mo. staff) Admin work day         April         Spring Break (CSEA mandatory vacation days)         SBAC Window Opens         May         SBAC Window Closes         Memorial Day (legal holiday)         June         Last day of School (last work day for 10 mo. staff)         Last work day for 11 mo. Classified staff	5 12 19 26 <i>M</i> 1 8 15 22 29 <i>M</i> 3	T           6           13           20           27           7           2           9           16           23           30           7           4	W           Z           14           21           28           APRIL           W           3           10           17           24           JUNE           W           5	Th         1         2         22         29         Th         4         11         18         25	2 <u>9</u> 16 23 F 5 12 19 26 F 7
1 8 15 22 29 <i>M</i> 4 11 18 25 <i>M</i> 6	T 2 9 16 23 30 T 7 5 12 19 26 7 7	W           3           10           17           24           31           WMARC           W           6           13           20           27           MAY           %           1           8	Th           4           11           18           25           Th           7           14           21           28           Th           2           9	5 12 19 26 <i>F</i> 1 8 15 22 29 <i>F</i> 3 10	1 15 5-9 19 20-22 23 15 8-12 22 22 24 27 27 6 13 19	January         New Years         Martin Luther King Jr.'s Birthday (legal holiday)         February         Intervention Conferences         Washington's Birthday (legal holiday) observed         Presidents' Week break (non-work days - 10 mo. staff)         Lincoh's Birthday (legal holiday) observed         March         Optional Professional Development Day (non-work day for 10 & 11 mo. staff) Admin work day         April         Spring Break (CSEA mandatory vacation days)         SBAC Window Opens         May         SBAC Window Closes         Memorial Day (legal holiday)         June         Last day of School (last work day for 10 mo. staff)         Last work day for 11 mo. classified staff         Juneteenth (legal holiday)	5 12 19 26 <i>M</i> 1 1 8 15 22 29 <i>M</i> 3 10	T           6           13           20           27           7           9           16           23           30           7           4           11	W           Z           14           21           28           APRIL           W           3           10           17           24           JUNE           W           5           12	Th         1         2         22         29         Th         4         11         18         25         Th         6         13	2 <u>9</u> 16 23 F 5 12 19 26 F 7 14
1 8 15 22 29 <i>M</i> 4 11 18 25 <i>M</i> 6 13	T           2           9           16           23           30           T           5           12           19           26           T           7           14	W           3           10           17           24           31           WARC           W           6           13           20           27           WAY           8           15	Th           4           11           18           25           Th           7           14           21           28           Th           2           9           16	5 12 19 26 7 8 1 8 15 22 29 7 8 8 15 22 29 7 7 8 10 10 17	1 15 5-9 19 20-22 23 15 8-12 22 24 27 27 6 13 19 25	January         New Years         Martin Luther King Jr. 's Birthday (legal holiday)         February         Intervention Conferences         Washington's Birthday (legal holiday) observed         Presidents' Week break (non-work days - 10 mo. staff)         Lincoln's Birthday (legal holiday) ob served         March         Optional Professional Development Day (non-work day for 10 & 11 mo. staff) Admin work day         Spring Break (CSEA mandatory vacation days)         SBAC Window Opens         May         SBAC Window Closes         Memorial Day (legal holiday)         June         Last day of School (last work day for 10 mo. staff)         Last day of School (last work day for 10 mo. staff)         Last work day for 11 mo. Classified staff         Juneteenth (legal holiday)         Last work day for 11 mo. Administrators (214 day work year)	5 12 19 26 <i>M</i> 1 1 8 15 22 29 <i>M</i> 3 10 17	Г           6           13           20           27           7           2           9           16           23           30           7           4           11           18	W           Z           14           21           28           APRIL           W           3           10           17           24           JUNE           W           5           12           19	Th         1         g         15         22         29         Th         4         11         18         25         Th         6         13         20	2 <u>9</u> 16 23 F 5 12 19 26 F 7 14 21
1 8 15 22 29 <i>M</i> 4 11 18 25 <i>M</i> 6	T 2 9 16 23 30 T 7 12 19 26 T 7	W           3           10           17           24           31           WMARC           W           6           13           20           27           MAY           8	Th           4           11           18           25           Th           7           14           21           28           Th           2           9	5 12 19 26 <i>F</i> 1 8 15 22 29 <i>F</i> 3 10	1 15 5-9 19 20-22 23 15 8-12 22 22 24 27 27 6 13 19	January         New Years         Martin Luther King Jr.'s Birthday (legal holiday)         February         Intervention Conferences         Washington's Birthday (legal holiday) observed         Presidents' Week break (non-work days - 10 mo. staff)         Lincoh's Birthday (legal holiday) observed         March         Optional Professional Development Day (non-work day for 10 & 11 mo. staff) Admin work day         April         Spring Break (CSEA mandatory vacation days)         SBAC Window Opens         May         SBAC Window Closes         Memorial Day (legal holiday)         June         Last day of School (last work day for 10 mo. staff)         Last work day for 11 mo. classified staff         Juneteenth (legal holiday)	5 12 19 26 <i>M</i> 1 1 8 15 22 29 <i>M</i> 3 10	T           6           13           20           27           7           9           16           23           30           7           4           11	W           Z           14           21           28           APRIL           W           3           10           17           24           JUNE           W           5           12	Th         1         2         22         29         Th         4         11         18         25         Th         6         13	2 <u>9</u> 16 23 F 5 12 19 26 F 7 14

Revised 11/28/22

## **APPENDIX C - PEER ASSISTANCE AND REVIEW PROGRAM**

#### 1. <u>Purpose</u>

- 1.1 The Peer Assistance and Review Program (Program) allows exemplary teachers to assist certain permanent and beginning teachers in the areas of subject matter knowledge, teaching strategies, and teaching methods.
- 1.2 The extent of the Program's assistance and review depends on whether the participating teacher is a beginning teacher, a volunteer permanent teacher, or a permanent teacher who has received an overall unsatisfactory evaluation in the areas of teaching methods and instruction. The Program's assistance shall be provided through Consulting Teachers as described in detail in Sections 4.2 and 4.3 of this document. This assistance shall not involve the participation in nor the conducting of the annual evaluation of certificated unit members as set forth in Article 7 of the Agreement and Education Code 44660, et seq., except for making available to the evaluator the results of some unit members' participation in the Program.
- 1.3 The Program resources shall be utilized in the following priority: first, for Participating Teachers with an unsatisfactory evaluation; second, for Beginning Teachers; third, for Voluntary Participating Teachers on evaluation cycle; and finally, for other Voluntary Participating Teachers.

## 2. <u>Definitions for Purposes of this Document</u>

2.1 <u>"Classroom Teacher" or "Teacher"</u>

Any member of the certificated bargaining unit who is covered by the certificated evaluation, Article 7 of the Agreement.

2.2 <u>"Participating Teacher"</u>

A unit member who is a classroom teacher who either volunteers or is required by this Agreement to participate in the Program.

2.3 <u>"Consulting Teacher"</u>

An exemplary teacher meeting the requirements of subsection 4.2.1 who is selected by the Joint Panel to provide Program assistance to a Participating Teacher.

## 2.4 <u>"Beginning Teacher"</u>

Any unit member having probationary or temporary status, or any District teaching intern participating in a program established according to Education Code Sections 44305, et seq. and 44325, et seq. This Peer Program is to be closely coordinated with other District programs for training and assistance to beginning teachers.

## 2.5 <u>"Voluntary Participating Teacher"</u>

Any unit member with permanent status whose last three annual performance evaluations have been rated as an overall "meets standards" and who selects the "Alternative Professional Growth Evaluation Plan" under Section 7.11 of the Agreement, and who elects to have a Consulting Teacher assist in that alternative evaluation plan; or any permanent teacher wanting to engage in a professional growth activity utilizing a Consulting Teacher's assistance.

## 2.6 <u>"Participating Teacher With An Unsatisfactory Evaluation"</u>

A unit member with permanent status whose most recent Formal Evaluation Summary contained a summary rating of "unsatisfactory," and an unsatisfactory rating in at least one of the following areas: instructional techniques and strategies; adherence to curricular objectives; or establishment and maintenance of suitable learning environment.

# 2.7 <u>"Principal" Or "Evaluating Principal"</u> The certificated administrator appointed by the District to evaluate a certificated teacher.

- 3. <u>Program Outline</u>
  - 3.1 For Participating Teachers With An Unsatisfactory Evaluation (Refer also to Diagram 1)
    - 3.1.1 Any permanent teacher with an overall unsatisfactory evaluation in the areas listed in Section 2.6 must participate in the Program.
    - 3.1.2 The Consulting Teacher's assistance and review shall focus on the specific areas recommended for improvement by the Participating

Teacher's evaluator after the Participating Teacher receives the unsatisfactory rating. (See Form 1.)

- 3.1.2.1 These recommendations shall be written, aligned with student learning, clearly stated, and consistent with Education Code Section 44662. These recommendations shall be considered as the performance goals required by Education Code Sections 44664(a) and 44500(b)(2).
- 3.1.2.2 The Principal and the Consulting Teacher assigned to the Participating Teacher shall meet and discuss the recommended areas of improvement outlined by the Principal and the types of assistance that should be provided by the Consulting Teacher.
- 3.1.2.3 The Consulting Teacher and the evaluating Principal are expected to establish a cooperative relationship and shall coordinate and align the assistance provided to the Participating Teacher.
- 3.1.2.4 The Consulting Teacher and the Participating Teacher shall meet to discuss the plan for assistance. After that meeting, the Consulting Teacher will provide the assistance set forth in Section 4.3, which shall also involve conducting multiple classroom observations of the Participating Teacher.
- 3.1.3 Before April 1 of each year, the Consulting Teacher shall complete a written report describing the teacher's participation in the Program consisting solely of: (1) the assistance provided to the Participating Teacher; and (2) the results of the assistance in the targeted areas. (See Form 2.) This report shall be submitted to the Joint Panel, with a copy also submitted to the Participating Teacher and the Principal.
- 3.1.4 The results of the teacher's participation in the Program shall be shared with the principal. While the Principal may consider the Consulting Teacher's written report, the Principal shall observe the teacher's

92

performance and prepare the evaluation required by Article 7 of the Collective Bargaining Agreement.

- 3.1.5 After receiving the report, the Joint Panel shall determine whether the Participating Teacher will benefit from continued participation in the Program.
- 3.1.6 The teacher will continue participating in the Program until the Joint Panel determines the teacher no longer benefits from participation in the Program, or the teacher receives a satisfactory evaluation, or the teacher is separated from the District. The District has the sole authority to determine whether the Participating Teacher has been able to demonstrate satisfactory improvement.
- 3.1.7 The Consulting Teacher shall conduct multiple classroom observations of each Participating Teacher.
- 3.1.8 The Consulting Teacher's report on the participation in the Program, as defined in subsection 3.1.3 above, shall be shared with the Principal. The Principal has the sole authority to determine whether the Participating Teacher has demonstrated satisfactory improvement.
- 3.1.9 To the extent required by law, the Joint Panel shall forward the names of the permanent teachers with unsatisfactory evaluations who, after sustained assistance, are unable to demonstrate satisfactory improvement.
- 3.2 <u>For Beginning Teachers</u> (Refer also to Diagram 2)
  - 3.2.1 A Consulting Teacher will be assigned to one or more Beginning Teachers to provide assistance. In the first year the Consulting Teacher shall concentrate the assistance in the area of the District's Teaching Standards. In the second year, the Consulting Teacher will focus the assistance in the areas listed by the evaluating Principal as needing improvement and/or assistance.

- 3.2.2 The Consulting Teacher and the evaluating Principal shall have a cooperative relationship, and shall coordinate the assistance provided to the Beginning Teachers.
- 3.2.3 Because Beginning Teacher participation in the Program is not legally mandated, neither the Consulting Teacher nor the Joint Panel will make written reports regarding individual Beginning Teachers, nor forward to the Board the names of individual Beginning Teachers who participated in the Program. The Consulting Teacher shall provide an annual assessment of the Program's overall effectiveness for Beginning Teachers and specific areas for improvement in the Program to the Joint Panel. The Joint Panel will annually report to the Governing Board and the Association on the overall effectiveness of the Program for Beginning Teachers.
- 3.3 <u>For Voluntary Participating Teachers</u> (Refer also to Diagram 3)
  - 3.3.1 Those teachers participating in an alternative evaluation plan set forth in Section 7.11 of the Agreement, and any teacher in the off year of the evaluation cycle, may utilize a Consulting Teacher's assistance.
  - 3.3.2 Voluntary Participating Teachers are expected to be high performing individuals who wish to grow and learn with the assistance from a peer, or who may be seeking assistance due to a change in assignment or the institution of new curriculum. The Program for Voluntary Participating Teachers will focus on practical application of certain teaching skills or the acquisition of a new subject matter. Each year the Joint Panel, in consultation with the District administration, may select certain curricular areas or skills for emphasis with a select number of Consulting Teachers.
  - 3.3.3 As required by the Section 7.11 of the Agreement, the volunteering teacher and the evaluating Principal must mutually agree to the teacher's participation in an alternative evaluation plan. If the plan is

approved by the Principal and involves a Consulting Teacher, the plan will be submitted to the Joint Panel for the assignment of a Consulting Teacher. The Consulting Teacher shall meet with the Principal and the Voluntary Participating Teacher for planning and coordinating the plan. The process for utilizing Consulting Teachers for professional growth on off-cycle years shall be the same as above, except that the procedures for evaluation contained in Article 7.11 of the Agreement need not be followed.

3.3.4 Because permanent teachers with satisfactory performance are not mandated by law to participate in the Program, neither the Consulting Teacher nor the Panel will make written reports regarding individual volunteer teacher participants, nor forward to the Board the names of volunteer teacher participants. Any reports will be made only as required by the individual plan. The Consulting Teacher shall provide to the Joint Panel an annual assessment of the Program's effectiveness for volunteer teacher participants and specific areas for improvement in the Program. The Joint Panel will annually report to the Governing Board and the Association on the overall effectiveness of the Program for volunteer teacher participants.

## 4. <u>Governance and Program Structure</u>

- 4.1 Joint Panel
  - 4.1.1 The Peer Assistance and Review Program will be administered by a Panel consisting of five members, three certificated classroom teachers appointed by CETA and two administrators appointed by the District. Qualifications for the teacher representatives shall be the same as those for Consulting Teachers as set forth in Section 4.2.1, A Panel member's term shall be three years, except the first terms of the teacher members will be one two-year term, one three-year term, and one four-year term.

- 4.1.2 The Joint Panel will make all decisions through consensus in the areas of appointments, reports and recommendations to the Governing Board, and Program plan and budget. Failing consensus, decisions will be made by majority vote. Four of the five Panel members will constitute a quorum for purposes of meeting and conducting business.
- 4.1.3 The Joint Panel's primary responsibilities involve establishing the annual Program and budget, and selecting and overseeing the Consulting Teachers. In addition the Panel is responsible for:
  - submitting to the Governing Board and the Association an annual evaluation of the Program's impact, including recommendations regarding Participating Teachers with unsatisfactory evaluations, and if necessary, forwarding names of individuals who, after sustained assistance, are unable to demonstrate satisfactory improvement;
  - assigning the Consulting Teachers;
  - reviewing Consulting Teachers' reports on Participating Teachers with permanent status referred to the Program because of unsatisfactory evaluations;
  - evaluating the effectiveness of the Consulting Teachers in the role of Consulting Teacher;
  - coordinating with the District to provide training for Consulting Teachers, for Panel members, and where appropriate, for Participating Teachers;
  - forwarding to the Human Resources Office at the end of the year all the records regarding the Program that shall be filed separately from the individual personnel records, except as set forth in section 3.1.8 in this document; and
  - establishing internal operating procedures and regulations necessary to carry out the requirements of the Education Code

96

and this document, including a procedure for selecting the Joint Panel's chair.

- 4.1.4 The Panel shall use the following procedure for establishing the annual Program plan and budget:
  - (a) By March 15 of each fiscal year the Panel will establish a Program and budget for the succeeding year, which will include:
    - the estimated state revenues for the Program;
    - the estimated expenditures, involving:
      - projected number of Participating Teachers;
      - projected (full and part-time) number of Consulting Teachers needed to service the projected need;
      - projected number of Consulting Teachers needed to service the projected need;
      - released time for the Panel and Consulting Teachers; pay for Panel members and Consulting Teachers that is consistent with the pay parameters established by the negotiating parties;
      - pay for Consulting Teachers; and
      - projected costs for training, administrative overhead, and if necessary, legal and consulting assistance.
  - (b) By April 1, the Program plan/budget will be submitted to the Association President and the Superintendent for approval. If the plan/budget is not approved by both parties, it may be modified by mutual agreement. By May 1, if the parties cannot reach agreement to either approve the plan/budget or to amend

it, then the plan/budget will be implemented as submitted by the Panel.

## 4.2 <u>Consulting Teachers</u>

- 4.2.1 <u>Minimum Qualifications for Consulting Teacher:</u>
  - a credentialed classroom teacher with permanent status and at least five years of recent teaching experience;
  - demonstrated exemplary teaching ability, as indicated by, among other things, effective communication skills, subject matter knowledge, knowledge and commitment to District curricular goals and standards, and mastery of a range of teaching strategies necessary to meet students' needs in different contexts;
  - ability to work cooperatively and effectively with other teachers and administrators, demonstrated effective leadership skills, and experience in working on school or District committees.
- 4.2.2 Consulting Teacher positions shall be filled by the District posting the position. Each applicant will be required to submit a completed application, which shall include at least three references from individuals who have direct knowledge of the applicant's abilities for the position. A reference from a Principal is preferred, although not necessary. All applications and references will be treated with confidentiality and will not be disclosed except as required by law. The Joint Panel will make the selection, which will be forwarded to the Superintendent. The Panel's procedures for selecting Consulting Teachers shall include provisions for classroom observation of Consulting Teacher candidates.
- 4.2.3 The Joint Panel will assign Consulting Teachers. Within the first six weeks of the regular school year, either the Consulting Teacher or the Participating Teacher may petition the Panel for an assignment change for good reasons. The Participating Teacher shall be allowed only one change per year.
- 4.2.4 A Consulting Teacher's term will be three years. To the extent that a Consulting Teacher is reassigned from his/her regular classroom teaching assignment to serve

as a Consulting Teacher, after completing service as a Consulting Teacher, the teacher shall be returned to a regular teaching assignment at the beginning of the following school year according to Article 8 of the Agreement. The District will attempt to return the teacher to a position similar to the one the teacher held before assuming the Consulting Teacher position.

- 4.3. Consulting Teachers shall provide assistance to Participating Teachers in the areas of subject matter knowledge, teaching strategies, and teaching methods. This assistance may include, but not be limited to, the following activities:
  - (a) providing consultative assistance to improve in the specific areas targeted by the evaluating Principal or the District Teaching Standards;
  - (b) meeting and consulting with the Principal or designee regarding the nature of the assistance being provided;
  - (c) observations of the Participating Teacher during periods of classroom instruction;
  - (d) allowing the Participating Teacher to observe the Consulting Teacher or other selected teachers;
  - (e) attending specific training in specified teaching techniques or in designated subject matter;
  - (f) demonstrating good practices to the Participating Teacher; and
  - (g) maintaining appropriate records of each Participating Teacher's activities and progress.
- 5. <u>Other Provisions</u>
  - 5.1 Functions performed by unit members under this document shall not constitute either management or supervisory functions as defined by Government Code Section 3540.1(g) and (m).
  - 5.2 Unit members who perform functions as Consulting Teachers or Panel members under this document shall have the same protection from liability and access to appropriate defense as other public school employees pursuant to Division 3.6 (commencing with Section 810) of Title 1 of the California Government Code.

## 5.3 <u>Records</u>

- 5.3.1 All documents and information relating to the participation in this Program will be regarded as a personnel matter and subject to the personnel record exemption of the California Public Records Act (Government Code Section 6250, et seq.). The annual evaluation of the Program's impact, excluding any information on identifiable individuals, shall be subject to disclosure under the Public Records Act.
- 5.3.2 All parts of the selection process of Consulting Teachers will be treated as confidential and will not be disclosed except as required by law.
- 5.3.3 All the documents for the Peer Program will be filed by the personnel office separately from the individual personnel records, except as set forth in 3.1.8 above.
#### FORM 1

# REFERRAL TO PEER ASSISTANCE AND REVIEW PROGRAM BASED ON UNSATISFACTORY EVALUATION AND RECOMMENDATIONS FOR IMPROVEMENT

Teacher:\_\_\_\_\_\_School Principal (or other evaluator):\_\_\_\_\_\_I am referring the teacher named above to the Peer Assistance and Review Program based on the teacher's Formal Evaluation Summary Rating of Unsatisfactory, as well as unsatisfactory evaluation in the following areas:



Instructional Techniques and Strategies



- Adherence to Curricular Objectives
- Establishment and Maintenance of Suitable Learning Environment

The teacher needs to improve in the specific areas described on the attached page(s) [*Principal to attach detailed description of areas in need of assistance*].

The assistance provided under the Peer Assistance and Review Program shall be designed to help the teacher to improve in the areas identified by the Principal or other evaluator.

Date

Signature of Principal Or Other Evaluator

#### FORM 2

#### DESCRIPTION OF PLANNED ASSISTANCE AND REPORT ON TEACHER'S PARTICIPATION IN PEER ASSISTANCE AND REVIEW PROGRAM FOR TEACHERS REFERRED BASED ON AN UNSATISFACTORY EVALUATION [To Be Completed By Consulting Teacher]

Participatin	g Teacher:	
Consulting	Teacher:	_
The Particip	pating Teacher was referred to the Peer Progr	am due to an unsatisfactory evaluation by
the Principa	al or other evaluator as determined by the rec	eipt of an average score of two (2) or less
in any one o	of sections A-D on the Certificated Employed	e Evaluation Form.
The Princip	al identified the areas in need of assistance of	the attached Referral To Peer Assistance
And Review	w Program (Form 1).	
The planne	d assistance and the results of the teacher's pa	articipation in the Program with reference
to each area	a needing assistance are described below: (A	ttach additional pages if necessary.)
1. Are	a in need of assistance:	
(1)	Brief description of planned assistance	:
(2)	<b>Results of Teacher's Participation</b> :	
	Planned Assistance Implemented	
	Planned Assistance Not Implemented	
Con	nments on implementation or failure to imple	ment planned assistance:
2. Are	a in need of assistance:	

(1)	Brief description of planned assistance:
(2)	Results of Teacher's Participation:
(_)	Planned Assistance Implemented
	Planned Assistance Not Implemented
	Comments on implementation or failure to implement planned assistance:
Are	a in need of assistance:
Are	a in need of assistance:
Are	a in need of assistance:
Are	a in need of assistance:
(1)	a in need of assistance:
(1)	Brief description of planned assistance:
(1)	Brief description of planned assistance:

## **Certification of Planned Assistance:**

The following signatures of the Consulting Teacher and Participating Teacher certify that the Consulting Teacher and the Participating teacher have met and discussed the planned assistance described above by the Consulting Teacher.

Date

Signature of Consulting Teacher

Date

Signature of Participating Teacher

## **Certification of Results:**

The Consulting Teacher's signature below certifies that the Consulting Teacher made the findings and comments regarding the Results of the Participating Teacher's participation in the Peer Assistance and Review Program indicated in the Results sections of this form.

Date

Signature of Consulting Teacher

The Participating Teacher's signature below certifies that the Participating Teacher has received a copy of the Consulting Teacher's findings and comments indicated in the Results sections of this form, and has been advised of his or her right to respond in writing to the report within one week of receipt of the completed report. The Participating Teacher's signature does not signify agreement with the Consulting Teacher's findings or comments.

Date

Signature of Participating Teacher

## APPENDIX D - EVALUATION FORMS (see following pages)

#### Campbell Union School District Permanent Certificated Pre-Conference and Evaluation Plan

Evaluatee						
Evaluator	School					
Pre-Conference and Evaluation Plan						
Pre-Conference and Evaluation Plan (must be completed by October 15)     Date:						
By mutual agreement the following three (3) Standards will be of focus:						
□ Standard 1	□ Standard 3	□ Standard 5				
□ Standard 2	Standard 4	Standard 6				
<u>Observations and Evaluation Summaries</u> □ Observation – 1 <sup>st</sup> Formal (complete by Jan	uary 15)	Date:				
<ul> <li>Interim Evaluation Summary Meeting (within <u>5</u> working days of observation)</li> </ul>	uary 13)	Date:				
□ Observation- 2 <sup>nd</sup> Formal (complete between	Date:					
□ Post-Observation Conference by May 3		Date:				
□ Formal Evaluation Summary (within <u>5</u> wo	rking days of Post Observation Conference)	Date:				

Reminder: Two Informal Observations will also be conducted within the evaluation timeline.

**Note:** If there are areas of concern that were not addressed on the Interim Evaluation Summary Form, please ensure that the employee is aware of the area in need of attention. (document date and time of discussion)

Updated 11/9/18

#### CAMPBELL UNION SCHOOL DISTRICT PRE-CONFERENCE EVALUATION PLAN

Teacher's Name:	School:	Grade/Subject:
Evaluator:	Date:	(to be completed by Oct. 15)
Check one:		
🗆 Temporary 🛛 🗆 Pro	bationary 2	
🗆 Probationary 1 🛛 Per	manent	
1. Engaging and Supporting All S	Students in Learning	
1.1 Using knowledge of students to enga	age them in learning	
1.2 Connecting learning to students' prio	r knowledge, background, life experience	s and interests
1.3 Connecting subject matter to meaning	ıgful real-life contexts	
	gies, resources, and technologies to meet	t students' diverse learning needs
1.5 Promoting critical thinking through in		
1.6 Monitoring student learning and adju	sting instruction while teaching	
Specific Focus:		
2 Creating and Maintaining Effe	ctive Environments for Student L	earning
	esponsibility within a caring community w	
respectfully		
constructive and productive interacti	The second	
2.3 Establishing and maintaining learning	genvironments that are physically, intelle	ctually, and emotionally safe
	nent with high expectations and appropria	
2.5 Developing, communicating, and ma	intaining high standards for individual and	d group behavior
<ol> <li>Employing classroom routines, proce students can learn</li> </ol>	dures, norms, and supports for positive b	ehavior to ensure a climate in which all
Specific Focus:		
-		
3. Understanding & Organizing S	Subject Matter for Student Learni	ing
<ol> <li>Demonstrating knowledge subject m resources</li> </ol>	atter, academic common core content sta	andards, and other state adopted
	opment and proficiencies to ensure stude	nt understanding of subject matter
3.3 Utilizing instructional strategies that a	2 2 2	are understanding of subject matter
	ologies, and standards-aligned instruction	al materials, including district adopted
materials, to make subject matter ac	cessible to all students	
3.5 Differentiating instruction to meet the access to the content	e needs of English learners and students	with special needs to provide equitable
Specific Focus:		

#### 4. Planning Instruction & Designing Learning Experiences for All Students

- 4. 1 Using knowledge of students' academic readiness, language proficiency, cultural background, and individual development to plan instruction
- 4.2 Establishing and articulating learning objectives for student achievement
- 4.3 Developing and sequencing long-term and short-term instructional plans to support student learning, such as pacing guides
- 4.4 Planning instruction that incorporates appropriate strategies to meet the learning needs of all students
- 4.5 Differentiation of instructional plans and curricular materials to meet the assessed learning needs of all students

Specific Focus:

#### 5. Assessing Student Learning

5.1 Applying knowledge of the continuum of assessments to assure student learning

- 5.2 Analyzing the assessment data collected from a variety of sources to inform instruction
- 5.3 Reviewing data, both individually and in collaboration with colleagues, to monitor student learning
- 5.4 Using assessment data to establish learning goals and to plan, differentiate, and modify instruction
- 5.5 Show evidence of student progress toward meeting grade level standards and established goals and outcomes
- 5.6 Involving all students in self-assessment, goal setting, and monitoring progress
- 5.7 Using available technologies to assist in assessment, analysis, and communication of student learning

5.8 Using assessment information to share timely and comprehensible feedback with students and their families **Specific Focus:** 

#### 6. Developing as a Professional Educator

- 6.1 Reflecting on teaching practice in support of student learning
- 6.2 Establishing professional goals and engaging in continuous and purposeful professional growth and development
- 6.3 Collaborating with colleagues and the broader professional community to support teacher and student learning
- 6.4 Working with families to support student learning
- 6.5 Completing non-instructional responsibilities such as adjunct and yard duty

6.6 Demonstrating professional conduct by participating fully, interacting respectfully, and maintaining integrity within the classroom and school community

**Specific Focus:** 

Unit member will use the Standard Evaluation Plan defined in Sections 7.1, 7.5, 7.7

Unit member will use the Alternate Evaluation Plan defined in section 7.11 (Permanent only)

Date

Unit member is responsible for implementing all teaching standards

Evaluator's Signature

Teacher's Signature

Page 2 of 2

Revised 9/9/11

#### CAMPBELL UNION SCHOOL DISTRICT CERTIFICATED FORMAL EVALUATION SUMMARY

Teacher's Name:	School:		Grade/S	ubjec	t:				
Evaluator:	Date:								
Check one     Probation       Temporary     Probation       Probationary 2     Permane		<ul><li>E = Exceeds Standards</li><li>PE = Progress Evident</li><li>U = Unsatisfactory</li></ul>				dent			
1. Engaging and Supporting All	Students in Learnin	g		E	MS	PE	NE*	U	
<ul> <li>1.1 Using knowledge of students to engage them in learning</li> <li>1.2 Connecting learning to students' prior knowledge, background, life experiences and interests</li> <li>1.3 Connecting subject matter to meaningful real-life contexts</li> <li>1.4 Using a variety of instructional strategies, resources, and technologies to meet students' diverse learning needs</li> <li>1.5 Promoting critical thinking through inquiry, problem solving, and reflection</li> <li>1.6 Monitoring student learning and adjusting instruction while teaching</li> <li>Comment:</li> </ul>									
2. Creating and Maintaining Eff	ective Environments	s for Student Learnin	g	E	MS	PE	NE*	U	
<ul> <li>2.1 Promoting social development ar respectfully</li> <li>2.2 Creating physical or virtual learn constructive and productive interacti</li> <li>2.3 Establishing and maintaining lear</li> <li>2.4 Creating a rigorous learning envi</li> <li>2.5 Developing, communicating, and</li> <li>2.6 Employing classroom routines, p students can learn</li> <li>Comment:</li> </ul>	ing environments that ons among students rning environments tha ironment with high exp I maintaining high stan	promote student learnin at are physically, intellec pectations and appropria dards for individual and	ng, reflect ctually, and ate support group beł	diver I emo t for a navio	sity, an otionally all stud	d enco / safe ents	ourage		
3. Understanding & Organizing	g Subject Matter for	Student Learning		E	MS	PE	NE*	U	
3.1 Domonstrating knowledge of sub	viact matter academic	common core content	tandarde						
<ul> <li>3.1 Demonstrating knowledge of subject matter, academic common core content standards, and other state adopted resources</li> <li>3.2 Applying knowledge of student development and proficiencies to ensure student understanding of subject matter</li> <li>3.3 Utilizing instructional strategies that are appropriate to subject matter</li> <li>3.4 Using and adapting resources, technologies, and standards-aligned instructional materials, including district adopted materials, to make subject matter accessible to all students</li> <li>3.5 Differentiating instruction to meet the needs of English learners and students with special needs to provide equitable access to the content</li> <li>Comment:</li> </ul>									

September 7, 2011

1 of 2

#### CAMPBELL UNION SCHOOL DISTRICT CERTIFICATED FORMAL EVALUATION SUMMARY

4. Planning Instruction and Designing Learning Experiences for All Students	E	MS	PE	NE*	U		
<ul> <li>4. 1 Using knowledge of students' academic readiness, language proficiency, cultural background, and individual development to plan instruction</li> <li>4.2 Establishing and articulating learning objectives for student achievement</li> <li>4.3 Developing and sequencing long-term and short-term instructional plans to support student learning, such as pacing guides</li> <li>4.4 Planning instruction that incorporates appropriate strategies to meet the learning needs of all students</li> <li>4.5 Differentiation of instructional plans and curricular materials to meet the assessed learning needs of all students</li> <li>Comment:</li> </ul>							
5. Assessing Student Learning	Е	MS	PE	NE*	U		
				Π			
<ul> <li>5.2 Analyzing the assessment data collected from a variety of sources to inform instruction</li> <li>5.3 Reviewing data, both individually and in collaboration with colleagues, to monitor student learning</li> <li>5.4 Using assessment data to establish learning goals and to plan, differentiate, and modify instruction</li> <li>5.5 Show evidence of student progress toward meeting grade level standards and established goals and outcomes</li> <li>5.6 Involving all students in self-assessment, goal setting, and monitoring progress</li> <li>5.7 Using available technologies to assist in assessment, analysis, and communication of student learning</li> <li>5.8 Using assessment information to share timely and comprehensible feedback with students and their families</li> <li>Comment:</li> </ul>							
6. Developing as a Professional Educator	E	MS	PE	NE*	U		
<ul> <li>6.1 Reflecting on teaching practice in support of student learning</li> <li>6.2 Establishing professional goals and engaging in continuous and purposeful professional growth and development</li> <li>6.3 Collaborating with colleagues and the broader professional community to support teacher and student learning</li> <li>6.4 Working with families to support student learning</li> <li>6.5 Completing non-instructional responsibilities such as adjunct and yard duty</li> <li>6.6 Demonstrating professional conduct by participating fully, interacting respectfully, and maintaining integrity within the classroom and school community</li> </ul>							
Formal Evaluation Summary Rating  Exceeds Standards Meets Standards Progress Evident Progress NE <sup>2</sup> *a rating of NE in <u>any category</u> requires that the Addendum to the Formal Evaluation Summary for A copy of this evaluation will be placed in your personnel file after 10 days. You may prepare a writte case, that response will be attached to this evaluation and placed in your file accordingly.	orm be		ed.		uch		

Evaluator's Signature	Date	
Teacher's Signature	Date	(Does not indicate agreement)

September 7, 2011

2 of 2

#### Campbell Union School District Temp/Prob Certificated Pre-Conference and Evaluation Plan

Evaluatee	Position_					
Evaluator	School	;				
Pre-Conference and Evaluation Plan	he completed by October 15)	Date:				
By mutual agreement the following three (3) Standards will be of focus:						
Standard 1	Standard 3	Standard 5				
□ Standard 2	□ Standard 4	□ Standard 6				
Observations and Evaluation Summaries						
$\Box$ Observation – 1 <sup>st</sup> Formal (must be completed	by November 2)	Date:				
<ul> <li>Post-Observation Conference and Interim (within <u>5</u> working days of observation)</li> </ul>	Evaluation Summary	Date:				
□ Observation- 2 <sup>nd</sup> Formal (must be completed	by February 1)	Date:				
□ Post-Observation Conference (within <u>5</u> worl	king days of observation)	Date:				
□ Formal Evaluation Summary (must be compl	eted by February 15)	Date:				

Reminder: Two Informal Observations will also be conducted prior to February 1.

**Note:** If there are areas of concern that were not addressed on the Interim Evaluation Summary Form, please ensure that the employee is aware of the area in need of attention. (document date and time of discussion)

Updated 9/29/20

#### CAMPBELL UNION SCHOOL DISTRICT CERTIFICATED (TEMPORARY & PROBATIONARY) INTERIM EVALUATION SUMMARY

Teacher's Name:	School:	G	irade/Su	Ibject				
Evaluator:	Date:							
	robationary 1 robationary 2	E = Exceeds StandardsMS = NPE = Progress EvidentNE = (PU = Unsatisfactory						
1. Engaging and Supporti	ng All Students in Learnin	ig	E	MS	PE	NE	U	
1.1 Using knowledge of students to engage them in learning         1.2 Connecting learning to students' prior knowledge, background, life experiences and interests         1.3 Connecting subject matter to meaningful real-life contexts         1.4 Using a variety of instructional strategies, resources, and technologies to meet students' diverse learning needs         1.5 Promoting critical thinking through inquiry, problem solving, and reflection         1.6 Monitoring student learning and adjusting instruction while teaching         Comment:								
2. Creating and Maintaini	ng Effective Environments	s for Student Learning	E	MS	PE	NE	U	
<ul><li>2.2 Creating physical or virtual leproductive interactions among st</li><li>2.3 Establishing and maintaining</li><li>2.4 Creating a rigorous learning</li><li>2.5 Developing, communicating,</li></ul>	earning environments that promo udents learning environments that are environment with high expectation and maintaining high standards	ng community where each student ote student learning, reflect diversity physically, intellectually, and emotic ons and appropriate support for all for individual and group behavior orts for positive behavior to ensure	v, and e onally sa students	ncourag ife s	e const	tructive	and	
3. Understanding & Org	anizing Subject Matter for	r Student Learning	E	MS	PE	NE	U	
<ul><li>3.2 Applying knowledge of stude</li><li>3.3 Utilizing instructional strateg</li><li>3.4 Using and adapting resource</li><li>make subject matter accessible to</li></ul>	nt development and proficiencies ies that are appropriate to subject s, technologies, and standards-a to all students	a core content standards, and other s to ensure student understanding o ct matter ligned instructional materials, includ ers and students with special needs	of subje ling dist	ct matte	er pted ma	aterials		

#### CAMPBELL UNION SCHOOL DISTRICT CERTIFICATED (TEMPORARY & PROBATIONARY) INTERIM EVALUATION SUMMARY

Teacher's Name:	School:	Gr	ade/Su	bject					
Evaluator:	ator: Date:								
Check one:	<ul> <li>Probationary 1</li> <li>Probationary 2</li> </ul>	E = Exceeds StandardsMS = MPE = Progress EvidentNE = (ProU = Unsatisfactory							
1. Engaging and Sup	porting All Students in Learnin	g	E	MS	PE	NE	U		
1.1 Using knowledge of students to engage them in learning         1.2 Connecting learning to students' prior knowledge, background, life experiences and interests         1.3 Connecting subject matter to meaningful real-life contexts         1.4 Using a variety of instructional strategies, resources, and technologies to meet students' diverse learning needs         1.5 Promoting critical thinking through inquiry, problem solving, and reflection         1.6 Monitoring student learning and adjusting instruction while teaching         Comment:									
2. Creating and Main	taining Effective Environments	s for Student Learning	E	MS	PE	NE	U		
<ul><li>2.2 Creating physical or vir productive interactions am</li><li>2.3 Establishing and maint</li><li>2.4 Creating a rigorous lea</li><li>2.5 Developing, communic</li></ul>	opment and responsibility within a cari tual learning environments that promo ong students aining learning environments that are p rning environment with high expectation ating, and maintaining high standards outines, procedures, norms, and suppo	bte student learning, reflect diversity, physically, intellectually, and emotion ons and appropriate support for all s for individual and group behavior	and en nally sa tudents	ncourag fe	e const	ructive	and		
3. Understanding 8	Corganizing Subject Matter for	<sup>•</sup> Student Learning	E	MS	PE	NE	U		
3.1 Demonstrating knowledge subject matter, academic common core content standards, and other state adopted resources 3.2 Applying knowledge of student development and proficiencies to ensure student understanding of subject matter 3.3 Utilizing instructional strategies that are appropriate to subject matter 3.4 Using and adapting resources, technologies, and standards-aligned instructional materials, including district adopted materials, to make subject matter accessible to all students 3.5 Differentiating instruction to meet the needs of English learners and students with special needs to provide equitable access to the content <b>Comment:</b>									

#### CAMPBELL UNION SCHOOL DISTRICT CERTIFICATED (PERMANENT) INTERIM EVALUATION SUMMARY

Teacher's Name:	School:		Gra	de/Su	bject:			
Evaluator:		Date:						
	E = Exceeds Standards MS NE = (Progress) Not Evident		PE = Progre	ss Evi	dent			
1. Engaging and Supporting All	Students in Learning			E	MS	PE	NE	U
1.1 Using knowledge of students to eng	age them in learning							
1.2 Connecting learning to students' pri-	In the second seco	experiences and inte	erests					
1.3 Connecting subject matter to meani		experiences and ma	510505					
1.4 Using a variety of instructional strat		gies to meet students	s' diverse lea	arnina	needs			
1.5 Promoting critical thinking through i	and the second			······9				
1.6 Monitoring student learning and adj								
Administrator:	Teacher:	2	Next Steps					
2. Creating and Maintaining Eff	ective Environments for	<b>Student Learnin</b>	g	Е	MS	PE	NE	U
2.1 Promoting social development and r								
productive interactions among st 2.3 Establishing and maintaining learnin 2.4 Creating a rigorous learning environ 2.5 Developing, communicating, and ma 2.6 Employing classroom routines, proc learn	g environments that are physic ment with high expectations a aintaining high standards for in	nd appropriate suppo dividual and group b	ort for all stu ehavior	idents limate		ch all st	cudents	can
	Teacher.		Next Steps					
1								
3. Understanding & Organizin	g Subject Matter for Stu	dent Learning		Ε	MS	PE	NE	U
			6					
3.1 Demonstrating knowledge subject n	aatter, academic common core	content standards	and other st		opted r			
<ul> <li>3.2 Applying knowledge of student deve</li> <li>3.3 Utilizing instructional strategies that</li> <li>3.4 Using and adapting resources, techn make subject matter accessible to</li> <li>3.5 Differentiating instruction to meet th content</li> </ul>	elopment and proficiencies to e are appropriate to subject ma nologies, and standards-aligned o all students	nsure student unders tter d instructional materi	standing of a	subjec g dist	rict adoj	r oted ma	aterials,	
Administrator:								
	Teacher:		Next Step					
	Teacher:		Next Step	:				
4. Planning Instruction and De		nces for All Stud					LIP	
4. Planning Instruction and De		nces for All Stud		E	MS	PE	NE	U
-	signing Learning Experie		lents	E				
4. 1 Using knowledge of students' acade	signing Learning Experie		lents	E				
4. 1 Using knowledge of students' acade instruction	signing Learning Experie	ciency, cultural back <u>o</u>	lents	E				
4. 1 Using knowledge of students' acade	signing Learning Experie emic readiness, language profic g objectives for student achiev	ciency, cultural back <u>c</u> ement	lents ground, and	E indivi	dual dev	/elopm	ent to p	

CA	MPBELL UNIC	ON SCHOO	L DISTRICT	
CERTIFICATED (	PERMANENT	) INTERIM	<b>EVALUATION</b>	SUMMARY

Administrator:	Teacher:		Next Steps	teps:				
5. Assessing Student Learn	ing			E	MS	PE	NE	ι
5.1 Applying knowledge of the cor	ntinuum of assessments to assure studer	t learning						
.2 Analyzing the assessment data	a collected from a variety of sources to int	form instruction						
.3 Reviewing data, both individua	Ily and in collaboration with colleagues, t	o monitor stude	ent learning					
.4 Using assessment data to esta	blish learning goals and to plan, different	iate, and modify	instruction					
.5 Show evidence of student prog	gress toward meeting grade level standar	ds and establish	ned goals an	d out	comes			
.6 Involving all students in self-a	ssessment, goal setting, and monitoring p	progress						
.7 Using available technologies to	assist in assessment, analysis, and com	nunication of st	udent learn	ing				
.8 Using assessment information	to share timely and comprehensible feed	back with stude	nts and thei	r fam	ilies			
Administrator:	Teacher:		Next Step	aps:				
5. Developing as a Profess	ional Educator			Е	MS	PE	NE	
				E	M2	PC	INE	
.1 Reflecting on teaching practice	in support of student learning							-
.2 Establishing professional goals	and engaging in continuous and purpose	ful professional	growth and	l deve	lopmen	t		
.3 Collaborating with colleagues a	and the broader professional community t	o support teach	ner and stud	ent le	arning			
A Marking with familias to supp	ort student learning							
. T WORKING WITH TAITINGS TO Suppt	esponsibilities such as adjunct and yard o	uty						
-			maintaining	intea	rity with	in the o	classroo	m
5.5 Completing non-instructional r	nduct by participating fully, interacting re	spectfully, and ı	manntainnig					
5.5 Completing non-instructional r 5.6 Demonstrating professional co		spectfully, and i	Next Step:					

Evaluator's Signature

Date

Teacher's Signature (Does not indicate agreement)

# ALTERNATIVE Professional Growth EVALUATION

## ALTERNATIVE PROFESSIONAL GROWTH EVALUATION

## <u> WHY:</u>

Research tells us that most successful teaching and learning experiences are done through self-assessment and evaluation. Students and adults both learn and grow best as self-evaluators.

In keeping with the research, it is important to develop an evaluation system that represents self evaluation with coaching support.

Collaboration among administration and staff is an important support to teacher evaluation.

Enhancing professional growth for teachers is one goal of evaluation, enhanced student performance being the primary goal.

## <u>WHO</u>:

The following personnel will be eligible to participate in an alternative evaluation system:

- \* All tenured certificated personnel
- \* Participation will be voluntary by teachers with the approval of site administrators
- \* There will be no limit on the number of participants at each site.

## <u>WHAT</u>:

Teachers will develop an action research question and collect evidence through a portfolio to address the research question. The plan must be based on teacher investigation, data collection and analysis and/or research.

## <u>WHEN</u>:

Timelines follow those in Article 7 of the Collective Bargaining Agreement.

\* Oct. 1: Information to unit members related to the evaluation process (Section 7.4)

\* Oct. 15: Develop Teacher Portfolio Research Question (Section 7.5)

\* May 1: Complete Final Reflections and present portfolio (Section 7.7.1)

## <u> HOW</u>:

- 1. Unit member develops a research question with administrative input.
- 2. Strategies and activities are identified by the unit member.
- 3. Expected outcomes are identified by the unit member.
- 4. Portfolio evidence will be self identified for collection throughout the school year.
- 5. Needed resources will be agreed upon and provided.
- 6. A timeline will be developed.

7. All plans must include the areas of evaluation listed in Section 7.2 of the Collective Agreement.

## ACTION RESEARCH: TEACHING DOMAINS

DOMAIN 1 Teachers Will Create & Maintain an Effective Environment for Student Learning (7.2 #4)	The teacher creates and maintains a smoothly functioning, safe learning environment in which students assume responsibility for themselves and one another, participate in decision-making, work collaboratively and independently, are treated fairly and respectfully, and engage in purposeful learning activities. Student expectations for behavior are clearly established, understood and consistently maintained. The physical environment and the arrangements within that environment support positive social interactions and facilitate equitable engagement of students in productive tasks.
DOMAIN 2	The teacher exhibits strong working knowledge of subject matter content
Teachers Will Und-	to be taught, the range of resources available for making content accessible,
erstand & Organize	and is able to use that knowledge to construct meaningful learning
Knowledge for	activities, products and long-term projects for all students. Students
Student Learning	understand the central themes and concepts within a content area and are
(7.2 #3)	able to apply those learnings to other content.
DOMAIN 3	The teacher plans instruction based upon knowledge of subject matter,
Teachers Will Plan	students and community. The richness of student diversities is reflected in
Instruction & Design	the planning process. Challenging but realistic goals are established for all
Learning Experiences	students, and instructional plans are designed to help students reach those
(7.2 #2/3)	goals.
DOMAIN 4 Teachers Will Engage All Students in Powerful Learning (7.2 #2)	The teacher builds on knowledge of students, subject matter, teaching and learning to enact effective and powerful learning opportunities that engage all students. A variety of teaching strategies and approaches are used so that all students understand and adapt learning goals as their own. Students become self-directed and are able to demonstrate what has been learned in meaningful and authentic ways.
DOMAIN 5 Teachers Will Assess Student Learning (7.2 #1)	The teacher establishes and clearly communicates performance expectations and collects information about student performance from a variety of sources. The teacher shares that information with students, parents and support personnel in ways that improve understanding, foster continued growth and development. The teacher adjusts instruction based on ongoing assessments to ensure that learning opportunities for students are enhanced and that personal learning is facilitated.
DOMAIN 6	The teacher reflects on his/her professional development while interacting
Teachers Will Develop	in a wider learning environment with students, other educators, parents
as Professionals to	and community members. The teacher recognizes his/her own
Improve Teaching &	responsibilities to a broader professional community and seeks to learn
Learning	from and contribute to that community to improve teaching and learning
(7.2 #5)	for all students.

*NOTE: Domains are based on the Draft Framework of expectations for Beginning Teachers in California developed by CTC/CDE and the BTSA program. These domains are under revision.* 

## TEACHING IN CALIFORNIA EXPECTATIONS FOR TEACHER DEVELOPMENT

DOMAIN 1: CREATE AND MAINTAIN AN EFFECTIVE ENVIRONMENT FOR STUDENT LEARNING	DOMAIN 2: UNDERSTANDING AND ORGANIZE CONTENT KNOWLEDGE FOR STUDENT LEARNING
Organizing the physical environment	Demonstrating knowledge of subject matter content
Planning and implementing classroom procedures and routines	Demonstrating understanding of key subject matter concepts and their interrelationships for students
Establishing and maintaining standards for student behavior	Using materials, resources and technologies to make subject matter accessible to students
Using instructional time effectively	Engaging students in learning experiences to increase understanding of content
Establishing a climate that promotes fairness and respect	
Promoting social development and group responsibility	Integrating ideas and information
DOMAIN 3: PLAN INSTRUCTION AND DESIGN LEARNING EXPERIENCES FOR ALL STUDENTS	DOMAIN 4: ENGAGE ALL STUDENTS IN POWERFUL LEARNING
Establishing and articulating goals for student learning	Making learning goals and instructional procedures clear to all students
Demonstrating understanding of students' developmental learning needs	Making content and processes comprehensible and meaningful to all students
Having knowledge of students' background and experience to plan instruction	Identifying and addressing multiple backgrounds of all learners
Valuing and drawing upon student diversity in planning instruction	Encouraging students to extend their thinking
Developing and sequencing instructional activities and materials for student learning	Fostering self-directed learning
Modifying instructional plans to adjust for student needs	Guiding and coaching students toward powerful learning experiences
DOMAIN 5: ASSESS STUDENT LEARNING	DOMAIN 6: DEVELOP AS A PROFESSIONAL EDUCATOR
Establishing appropriate learning expectations for all students	Reflecting on teaching and learning
Collecting and using multiple sources of information about students and their work	Involving families and community members in student learning
Involving students in assessing their own work	Working with colleagues to improve teaching and learning
Communicating with students, parents and other audiences about student progress	Actively pursuing opportunities to contribute and grow professionally
Using the results of assessments to plan for future instruction	

#### TEACHER PORTFOLIO ACTION PLAN ASSESSMENT WORKSHEET

#### **<u>Quick Write</u>** (Unit Member/Administrator directed)

With your administrator review investigation data and brainstorm what area of teaching you would like to focus on in the portfolio process. Use the Summary Worksheet and your classroom experience to help you choose a target area. Sketch out your ideas in the space below.

#### **Developing your Action Research Question** (Unit Member directed)

Determine which part of your brainstorm sketch is most important to the students in your classroom and to your practice. Prioritize your thoughts below according to the six teaching Domains.

1.

2.

3.

Choose one area, based on the Domains, to focus on.

1.

Why is this area important to you and your students?

What do you need to do to investigate this area?

#### Writing an Action Research Question (Unit Member Directed)

Turn your brainstorming notes into a question that focuses on one domain area of teaching. The question will be meaningful to your classroom and relevant to your development as a teacher. The question needs to be broad so that you can spend several months thinking, learning and growing in the area as a professional. Write your question below.

#### **Support for Action Research** (Unit Member and Support Provider Directed)

Plan with your support provider how you will investigate your question and what kinds of support you will need to feel successful. Take notes below as you discuss your action research question and how you will work together to accomplish the teacher portfolio.

## TEACHER PORTFOLIO RESEARCH QUESTION

Date:	
Teacher:	School:
Research Question:	

Teacher's Signature:\_\_\_\_\_

Administrator's Signature:\_\_\_\_\_

Strategies and Activities	Outcomes	Resources Needed	Portfolio Evidence	Timeline
What do I need to do to address my research question?	How will what I do benefit my students?	Time, materials, training, information, data, personnel	What evidence will I need to indicate that I have reached my outcomes?	

<b>Strategies and Activities</b> What do I need to do to address my research question?	<b>Outcomes</b> How will what I do benefit my students?	<b>Resources Needed</b> Time, materials, training, information, data, personnel	<b>Portfolio Evidence</b> What evidence will I need to indicate that I have reached my outcomes?	Timeline
Administrator's Comments:				

## WHOLE PORTFOLIO RUBRIC: DEVELOPING AS A PROFESSIONAL

DEVELOPING	MATURING	ACCOMPLISHED
Portfolio provides evidence that the unit members participated in inquiry-based learning about teaching through the development of a research question, collection and analysis of evidence related to the question.	Portfolio provides clear evidence that the unit member used inquiry-based research to increase the effectiveness of classroom practices.	Portfolio provides substantial evidence that the unit member analyzed data from inquiry based research and used findings to develop classroom practices.
Portfolio provides evidence that the unit member grew professionally, e.g. reflection about specific lessons, evidence of participating in professional growth activities related to professional growth goals.	Portfolio provides clear evidence that the unit member grew professionally, e.g., reflection about professional successes, challenges and planning for the future; strategic use of curriculum, instruction and assessment strategies to meet the needs of all students.	Portfolio provides substantial evidence that the unit member grew professionally, e.g., reflection across evidence leading to new understanding and improved practices; contributing or applying knowledge and understanding about teaching, learning and assessment in ways that benefit students, communities and the profession.
Portfolio provides evidence about knowledge of students' experience and backgrounds.	Portfolio provides clear evidence of involving families or community members in student learning.	Portfolio provides substantial evidence of systematically involving families and community members as colleagues in student learning.
Portfolio provides evidence of involvement in a larger educational community.	Portfolio provides clear evidence about unit members' utilization of the knowledge gained from the larger educational community to improve practice.	Portfolio provides substantial evidence of unit members involvement in the larger educational community in a variety of ways that integrate professional practice.

#### PORTFOLIO REVIEW AND REFLECTION FORM

Use the whole portfolio rubric and reflect on the following

- \* What are your research findings (identify 3-5)?
- \* What evidence best documents your findings?
- \* How did your students benefit from your investigation?

Use the whole portfolio Rubric, review the whole portfolio, taking notes about specific evidence for the following:

- \* How does your portfolio show that you grew professionally?
- \* How does your portfolio show that you reflected about classroom practices?
- \* How does your portfolio show that your teaching addressed experiences and backgrounds of students?
- \* How does your portfolio show that you were involved in a larger educational community?

## ALTERNATIVE PROFESSIONAL GROWTH EVALUATION

## FINAL REFLECTIONS: GUIDELINES FOR PORTFOLIO PRESENTATION

Bargaining Unit Member:

**Self Reflection**: a brief statement about the portfolio, including what it demonstrates about teaching, and what the teacher learned about his or herself from the portfolio process.

**<u>Significant Findings</u>**: a brief overview of the research question a 3-5 significant findings about practice.

**<u>Change in Practice</u>**: teacher then discusses an entry/evidence that highlights changes in practice in a meaningful, unusual or thoughtful way.

**<u>Planning for Future</u>**: based on the portfolio rubric, and any relevant conversation guide and domain, teacher assesses areas of strength and plans for next steps in professional development.

Final Reflections:

Administrator's comments:

## **APPENDIX E - SHARED TEACHING CONTRACT GUIDELINES**

## SHARED TEACHING CONTRACT GUIDELINES

- Teachers who desire to share a contract are responsible for selecting a qualified and compatible partner.
- Together, the teachers will develop their plan for shared a contract including the details of responsibilities, percent of contract, etc.
- Teachers will complete the "Application for Shared Contract".
- Teachers will meet with the site principal to discuss their proposed plan.
- The principal will review the application and plan and submit his/her recommendation to the Human Resources supervisor.

## **Qualifications for Teachers Requesting Shared Contract**

#### **Required**

- 1. Approval from the site Principal or department lead (for Special Education classes).
- 2. At least one of the partners must have permanent status as a certificated employee in Campbell Union School District (CUSD) in a related grade or area.
- 3. At least one of the partners must have a minimum of two (2) years of experience at a similar grade level or its equivalent.
- 4. The permanent employee(s) must have received three (3) consecutive years of satisfactory evaluations.
- 5. Applicants must have demonstrated skills in organization communication and collaboration.
- 6. Knowledge of and demonstrated ability to differentiate instruction to meet the needs of all learners.

#### **Preferred**

- 1. Familiarity with collegial support methods (teaming, peer coaching, problem solving, etc.).
- 2. Demonstrated leadership capabilities.
- 3. Partners have worked together previously at the same location or on a project.

#### SHARED TEACHING CONTRACT GUIDELINES

## **Instructions Regarding Application for Shared Contract**

- 1. Completion of the "Application for Shared Contract" is the responsibility of the teachers.
- 2. Application must include the following:
  - a. Principal's signature and recommendation.
  - b. Full Time Equivalent (FTE) designation (i.e., what percentage of an FTE contract will each partner fulfill, and for permanent or probationary employees, the percentage of an FTE that the employee will be considered to be on unpaid leave).
  - c. Specific teaching schedule for each partner to exchange information, plan and collaborate with colleague and communicate with parents.
  - d. How the will meet student programmatic site and district needs.
  - e. Responsibility of each partner for supervision duties, preparation of progress reports/report cards, staff meetings, Back-to-School nights, Open House, PTA/School Site Council meetings, opening and closing of the school year, and adjunct duties (instructional and non-instructional).
  - f. Adjunct school and district commitments made by each partner.
  - g. Written commitment by both parties to fulfill all responsibilities and duties related to the instructional program in the event of a long-term absence or illness by one of the partners.
- 3. Selection Criteria
  - a. The following criteria shall be considered should there be a situation in which there are more qualified applicants for shared positions than there are shared contract positions available.
    - 1. The applicants meet all required qualifications.
    - 2. New qualified applicants will receive priority over those who have already shared contracts at a site during the three (3) previous school years.
    - 3. Date that the application was approved by the Superintendent or Designee.

**Application for Shared Contract** 

SCHOOL:	SCHOOL YEAR:
TEACHER:	
EMPLOYEE STATUS: (check one)	$\Box$ PROBATIONARY $\Box$
Contract FTE Requested%	
Commensurate FTE Leave Requested%	
Grade Level Requested	
ExperienceGrade Level(s) School	
TEACHER:	
EMPLOYEE STATUS: (check one)	$\Box$ PROBATIONARY $\Box$
Contract FTE Requested%	
Commensurate FTE Leave Requested%	
Grade Level Requested	
ExperienceGrade Level(s) School	l(s)
Grade Level(S) School	1(5)

ATTACH REQUIRED DOCUMENTS TO THIS PAGE.

#### **Application for Shared Contract**

## A. RATIONALE FOR REQUEST

Attach a brief rationale for your request. Please be as specific as possible regarding advantages for the students, instructional program and school.

#### **B. TEACHING SCHEDULE**

Attach a brief description of each teacher's responsibilities and/or subjects and days to be taught for the school year. Instructional schedule should provide for consistent teacher delivering specific content, i.e. math being taught daily by the same teacher.

#### C. CONDITIONS OF AGREEMENT

- 1. Both teachers are to be available for joint parent conferences and will accommodate any requests for such. Special events, such as, Back to School, Open House, music programs, etc., will be the responsibility of both teachers to be arranged to the mutual satisfaction of both teachers and the principal.
- 2. Evaluation for the content areas will be consistent with the Collective Bargaining Agreement. Each teacher will be responsible for his/her own curriculum plans, other activities and special projects, i.e., science projects, assessment, computers.
- 3. A management plan for handling students, including daily routines will be outlined for the students and parents. A mutually agreed upon set of expectations and consequences will be the standard for student behavior. This plan will address all aspects of student management and will be distributed to students, parents and the principal.
- 4. During times when one (1) teacher in a shared contract is absent for any reason, the other teacher sharing the contract will be allowed to substitute for shared contract partner only.
- 5. Each of the above teachers will receive prorated sick leave and other mandated benefits based upon <u>one</u> FTE.

6. Each unit member working less than seventy-five percent (75%) shall accrue service credit for step advancement in direct relation to their percentage of employment. No unit member shall remain more than two (2) years on a single, existing, salary schedule step. Any unit member working seventy-five percent (75%) or more of any given school year shall receive a full year's service credit for each year worked.

#### Shared Contract Agreement

- 1. Educational advancement (accumulated units) will remain the same as the current CUSD/CETA Agreement.
- 2. The Shared Contract plan is to be viewed as a one-year request. Continuation of the Shared Contract plan shall require a new request each year.
- 3. During the first weeks of school, the above teachers will hold a special meeting with parents to explain, a) the Shared Contract plan, and b) methods for parents to communicate with each teacher as the need arises.
- 4. All meetings concerning the relationship of the Shared Contract partnership will be attended by *both* teachers. Any informational meetings (i.e., staff, scheduling, etc.) will be attended by the partner working that specific day. It is the responsibility of the partners to keep each other informed. All decisions made concerning schedules, events, etc., will be made jointly and with concern for both partner's personal feelings, bias, and educational beliefs.
- 5. Both teachers will be available for parent conferences by using a rotating schedule.
- 6. During the effective dates of this agreement, permanent and probationary certificated employees sharing a contract are considered to be on unpaid leave from that portion of an FTE contract which the employee is sharing with another certificated employee.

We understand and agree with the conditions stated above. We accept these as conditions of employment under this Shared Contract Agreement.

Teacher (print)	Teacher signature	Date
Teacher (print)	Teacher signature	Date
Principal (print)	Principal signature	Date
Superintendent / Designee (print)	Superintendent/Designee signature	Date